DECLARATION

of

PROTECTIVE COVENANTS

and

OWNERS ASSOCIATION

for

COOL SPRINGS

EAST SIDE

Prepared by:
Hines Interests Limited Partnership
Development Manager
for
Cool Springs Real Estate Associates, L.P.

October 4, 1994

Recorded Inis	nay or	1994
	At	
Deedbook	, Pages te Tennessee, Real P	oroperty Records

-

TABLE OF CONTENTS

ARTICLE I

GENERAL

	Project Area	1
1.1	Danier in Danier in the Control of t	
1.2	General Statement	1
1.3	General Statement	
	ARTICLE II	
	,	
	DEFINITIONS	
	Administrative Fundions	2
2.1	Annexable Area	2
2.2	Annexed Property	2
2.3	Applicant	2
2.4	Appointment Period	2
2.5	Appointment Penod	2
2.6	A788	2
2.7	Articles of Incorporation	7
2.8	Assessment	7
2.8	Assessment Year	3
2.10	Board	ব
2 11	Budget	3
2.12	Bylaws	3
2.13	Class & Owner	3
2.14	Common Assessment	3
2.15	County Records	3
2.15	Doctarant	. 4
2.17	Declaration	4
2.18	Delegate	7
2.19	Delegate Delegate District	7
2.20	Development Review Committee	. 4
2.21	Development Review Committee Representative	
2.22	First Mortgage	. 4
2.23	First Mortgagee	4
2.24	First Subdivision	
2.25 2.26	Foreclosure	. 4
	Improvement to Property	(
2.27 2.28	Improvements	4
2.29	Master Associates	(
2.25	Mester Association Area	
2.31	Master Association Properties	!
.2.32	Mortgage	!
2.33	Молаган	
2.34	Mondagor	
2.35	Notice and Hearing	***
2.38	Notice of Completion	•••

TABLE OF CONTENTS

1	Notice of Noncompliance	5
	Notice of Withdrawal	5
	OwnerPerson	⊋ ₽
	Record. Recorded or Recordation	D A
֡	Reimbursoment Assessment	R
	Related User	
	Rules and Regulations	6
	Site	6
	Special Assessment	6
	Subassociation	6
	Supplemental Common Assessment	6
	Supplemental Declaration	8
	Utility Easement Area	
	Voting Unit	1
	ARTICLE III	
	ANNEXATION TO MASTER ASSOCIATION AREA	
	First Subdivision Made Subject To Doctaration	7
	Property Which May Be Annexed	7
	Manner Of Annexation	7
	Withdrawal Of Annexed Property By Declarant	8
	Expansion Of Annexable Area	8
	ARTICLE IV	
	MASTER ASSOCIATION OPERATIONS	
	Master Association	9
	Board	ĝ
	Board	8
	Board	8 8 8
	Board	8 8 8
	Board	9 9 9 9
	Board	8 8 8
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V	9 9 9 9
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION	9 9 9 10 10
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION General Duties And Powers Of Master Association	9 9 9 11 10
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION General Duties And Powers Of Master Association Duty To Accept Property And Facilities Transferred By Declarant	9 9 9 11 11
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION General Duties And Powers Of Master Association Duty To Accept Property And Facilities Transferred By Declarant Duty To Manage And Care For Property	9 9 9 11 1 1 1 1 1
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION General Duties And Powers Of Master Association Duty To Accept Property And Facilities Transferred By Declarant Duty To Manage And Care For Property Duty To Pay Taxes	99911
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION General Duties And Powers Of Master Association Duty To Accept Property And Facilities Transferred By Declarant Duty To Manage And Care For Property Duty To Pay Taxes Casualty Insurance	9 9 9 1 1 1 1 1 1 1 1 1 1
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION General Duties And Powers Of Master Association Duty To Accept Property And Facilities Transferred By Declarant Duty To Manage And Care For Property Duty To Pay Taxes Casualty Insurance Liability Insurance General Provisions Respecting Insurance	999911
	Board Membership in Master Association Classes Of Membership Establishment of Delegate Districts Voting Right Of Owners Voting Rights Of Delegates ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION General Duties And Powers Of Master Association Duty To Accept Property And Facilities Transferred By Declarant Duty To Manage And Care For Property Duty To Pay Taxes Casualty Insurance Liability Insurance Liability Insurance	999911111111111111111111111111111111111

TABLE OF CONTENTS		
	Duty To Prepare Budgets	
5.10		
5.11 5.12	Duty To Levy And Collect Assessments 12 Duty To Invest Funds 12	
5.12		
5.14		
5.15		
5.16	Ta Destida Sastica Of Strategical Landing Control of the Control o	
5.17	Power To Provide Special Services For Owners	
5.18	Power To Grant Easements	
5.18	Power To Convey And Dedicate Property to Government Agents of the Power To Borrow Money	
5.20	Power To Employ Managers: Management Contracts	
5.21	Power To Engage Employees, Agents and Consultants	
5.22	General Corporate Powers	
5.23	General Corporate Powers	
	ARTICLE VI	
	MASTER ASSOCIATION PROPERTIES	
6.1	Owner's Rights Of Use and Enjoyment Generally	
6.2	Disk of Mariae Association To Requiste Use	
6.3	Diable Of Marter Association To Allow Public Use	
6.4	the date of Marian Association Properties	
6.5	1 inhility Of Owner For Damage By Owner	
6.6	Company Contracts Of Required Improvements 10 Master	
	AIntian Deposition	
6.7	** - t - t - a - a - i - i - i - i - i - i - i - i	
5.8	Title To Master Association Properties On Dissolution Of Master Association	
	ARTICLE VII	
	DECLARANT'S RIGHTS AND RESERVATIONS	
7.1	Period Of Declarant's Rights And Reservations	
7.2	Dialy To Conduct Additional Improvements On Master Association	
	Propedies 10	
7.3	Declarant's Rights To Use Master Association Properties In Promotion and	
	Marketing Of Master Association Area	
7.4	Declarant's Rights To Complete Development Of Cool Springs	
7.5	Declarant's Rights To Grant and Create Easements	
7.6		
	ARTICLE VIII	
ASSESSMENTS		
8.1	Convenant To Pay	
8.2	A anual Rudgets And Financial Reports.	
8.3	l eving Of Common Assessments	
8.4	Amount Of Common Assessments	
8.5	Commencement Of Common Assessments.	
8.6	Payment Of Common Assessments	

	TABLE OF CONTENTS	
8.7	Failure To Establish Common Assessments	9
8.8	Supplemental Common Assessments	18
8.9	Special Assessments	8
8.10	Reimbursement Assessments	18
8.11	Diagle Of Line Eng Accessments	20
8.12	Effect Of Nonpayment Of Assessment; Remedies Of The Association	20
8.13	Exempt Property	20
8.14	Estoppel Certificate	20
8.15	No Offsets	21
8.13		
	ARTICLE IX	
	GENERAL RESTRICTIONS APPLICABLE TO PROPERTY	
9.1	Application Of Restrictions	2 1
9.2	Maintenance Of Property	21
9.3	Noxious Or Offensive Adlvity	22
9.4	Annoying Sounds Or Orders	22
9.5	Hazardous Activities	22
9.8	Garbage And Trash	22
9.7	Temporary Structures	22
9.8	Antennas, Aerials and Dishes	22
	Signs and Advertising Devices	22
9.9	Prohibited Uses	23
9.10	Maintenance Of Drainage	23
9.11	Compliance With Insurance Requirements	23
9.12	Compliance With I are	24
9.13	Compliance With Laws	24
9.14	Sewage Disposal Systems	24
9.15	Water SystemsRestoration In Events Of Damage Or Destruction	24
9.18	Restoration in Events of Daniage of Destruction	21
9.17	Land Coverage	24
9.18	Height Of Rooftop Structures	25
9.19	Parking Structures	29
9.20	Parking And Circulation	
9.21	Curb Cuts	
9.22	Pedestrian Circulation	
9.23	Loading And Servicing Areas	
9,24	Storage Areas	
9,25	Utilities	. 26
9.26	Landscape Development	. 26
	ARTICLE X	
	DEVELOPMENT APPROVAL	
10.1	Approval Of Improvements Required	. 27
10.2	Improvement To Property	27
10.3	Membership Of Committee	. 27
10.4	Address Of Committee	. 27
10.5	Submission Of Plans	. 27
10.6	Pre-design Conference	
10.7	Schematic Design Review	
10.8		
	•:-	

TABLE OF CONTENTS

	Construction Documents Review	
10.10	Changes After Final Review	
10.11	Criteria For Approval	29
10.12	Committee Guidelines Or Rules	29
10.13	Development Review Fee	30
10.14	Decisions Of Committee	30
10.15	Appeal To Board	30
10.16	Failure Of Committee To Act On Plans	30
10.17	Prosecution Of Work After Approval.	
10.18	Notice Of Completion.	
10.19	Inspection Of Work	21
10.18	Notice Of Noncompliance	
	Failure Of Committee To Act After Completion	
10.21	Pallure Of Committee 10 ACI After Completion	31
10.22	Appeal To Board Of Finding Of Noncompliance	
10.23	Correction Of Noncompliance	
10.24	No Implied Waiver Or Estoppel.	
10.25	Committee Power To Grant Variances	
10.26	Compensation Of Members	
10.27	Committee Representative	
10.28	Records Of Actions	
10.29	Estoppel Certificates.	33
10.30	Nonliability For Committee Action	
10.31	Construction Period Exception	33
10.32	Appeal To Court Of Law	33
10.33	Development By Declarant	
	ARTICLE XI MISCELLANEOUS PROVISIONS	
11.1		
11.2	Duration	2.1
	Duration	34
11.5	Land Outside Master Association Area	34
11.3 11.4	Land Outside Master Association Area	34 34
11.4	Land Outside Master Association Area. Amendment To Declaration By Declarant. Amendment To Declaration By Owners.	34 34 34
11.4 11.5	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information	34 34 34 35
11.4 11.5 11.6	Land Outside Master Association Area	34 34 34 35 35
11.4 11.5 11.6 11.7	Land Outside Master Association Area. Amendment To Declaration By Declarant. Amendment To Declaration By Owners. Association Right To Mortgage Information. Notices. Enforcement.	34 34 35 35
11.4 11.5 11.6 11.7 11.8	Land Outside Master Association Area. Amendment To Declaration By Declarant. Amendment To Declaration By Owners. Association Right To Mortgage Information. Notices. Enforcement. No Obligation To Enforce Declaration.	34 34 35 35 35 35
11.4 11.5 11.6 11.7 11.8 11.9	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nuisance	34 34 35 35 35 36
11.4 11.5 11.6 11.7 11.8 11.9	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nulsance Enforcement By Self-Help	34 34 35 35 35 38 38
11.4 11.5 11.6 11.7 11.8 11.9 11.10	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nulsance Enforcement By Self-Help Violation Of Law	34 34 35 35 36 38 38
11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nuisance Enforcement By Self-Help Violation Of Law No Waiver	34 34 35 35 36 36 38 38
11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nuisance Enforcement By Self-Help Violation Of Law No Waiver Remedies Cumulative	34 34 35 35 36 36 36 36
11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nulsance Enforcement By Self-Help Violation Of Law No Waiver Remedies Cumulative Costs And Attorney's Fees	34 34 35 35 36 38 38 38 38 38
11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nulsance Enforcement By Self-Help Violation Of Law No Waiver Remedies Cumulative Costs And Attorney's Fees Limitation On Liability:	34 34 35 35 38 38 38 38 38 38 38 38
11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nulsance Enforcement By Self-Help Violation Of Law No Waiver Remedies Cumulative Costs And Attomey's Fees Limitation On Liability: General Development Information	34 34 35 35 36 38 38 38 38 36 36 36 36 36
11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16	Land Outside Master Association Area. Amendment To Declaration By Declarant. Amendment To Declaration By Owners. Association Right To Mortgage Information. Notices. Enforcement. No Obligation To Enforce Declaration. Violations Constitute A Nuisance. Enforcement By Self-Help. Violation Of Law. No Waiver. Remedies Cumulative. Costs And Attomey's Fees. Limitation On Liability: General Development Information. Use of "Cool Springs" Name.	34 34 35 35 36 38 38 38 36 36 36 37 37
11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17	Land Outside Master Association Area. Amendment To Declaration By Declarant. Amendment To Declaration By Owners. Association Right To Mortgage Information. Notices. Enforcement. No Obligation To Enforce Declaration. Violations Constitute A Nuisance. Enforcement By Self-Help. Violation Of Law. No Waiver. Remedies Cumulative. Costs And Attomey's Fees. Limitation On Liability: General Development Information. Use of "Cool Springs" Name. Governing Law.	34 34 35 35 36 38 38 36 36 36 37 37
11.4 11.5 11.6 11.7 11.8 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19	Land Outside Master Association Area. Amendment To Declaration By Declarant. Amendment To Declaration By Owners. Association Right To Mortgage Information. Notices. Enforcement. No Obligation To Enforce Declaration. Violations Constitute A Nulsance. Enforcement By Self-Help. Violation Of Law. No Waiver Remedies Cumulative. Costs And Attomey's Fees. Limitation On Liability General Development Information. Use of "Cool Springs" Name. Governing Law. Captions For Convenience.	34 34 35 35 36 38 38 38 38 36 36 37 37 37
11.4 11.5 11.6 11.7 11.8 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nuisance Enforcement By Self-Help Violation Of Law No Waiver Remedies Cumulative Costs And Attomey's Fees Limitation On Liability: General Development Information Use of "Cool Springs" Name Governing Law Captions For Convenience Interpretation	34 34 35 35 36 38 38 38 38 36 37 37 37
11.4 11.5 11.6 11.7 11.8 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20 11.21	Land Outside Master Association Area Amendment To Declaration By Declarant Amendment To Declaration By Owners Association Right To Mortgage Information Notices Enforcement No Obligation To Enforce Declaration Violations Constitute A Nulsance Enforcement By Self-Help Violation Of Law No Waiver Remedies Cumulative Costs And Attomey's Fees Limitation On Liability: General Development Information Use of "Cool Springs" Name Governing Law Captions For Convenience Interpretation Conflicts In Legal Documents	34 34 35 35 36 38 38 38 36 36 37 37 37 37 37
11.4 11.5 11.6 11.7 11.8 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20	Land Outside Master Association Area. Amendment To Declaration By Declarant. Amendment To Declaration By Owners. Association Right To Mortgage Information. Notices. Enforcement. No Obligation To Enforce Declaration. Violations Constitute A Nulsance. Enforcement By Self-Help. Violation Of Law. No Waiver Remedies Cumulative. Costs And Attomey's Fees. Limitation On Liability General Development Information. Use of "Cool Springs" Name. Governing Law. Captions For Convenience.	34 34 35 35 36 38 38 38 36 36 37 37 37 37 37

BK 1235 PG 731

TABLE OF CONTENTS

11.23	Severability	37
11.24	Notice Of Sales, Lease Or Mortgage	38
	No Reverter	
	Effective Date Of Declaration	

DECLARATION OF PROTECTIVE COVENANTS AND OWNERS ASSOCIATION FOR COOL SPRINGS

EAST SIDE

This Master Declaration of Protective Covenants and Owners Association for Cool Springs East Side is hereby declared on October 4, 1994, by Cool Springs Real Estate Associates, L.P., a Tennessee limited partnership ("Declarant"), so that the covenants, conditions, restrictions, easements, reservations, rights-of-way, equitable servitudes and other provisions stated in this Declaration shall run with the land and shall be binding upon the real property in Williamson County, Tennessee described in Exhibit "A" and upon any parcels of property subsequently annexed hereto in accordance with the provisions of this Declaration and subsequently referred to as "Cool Springs," and shall be binding upon and inure to the benefit of all parties having any right, title, or Interest in such real property and parcels and their heirs, personal and legal representatives, successors and assigns. Declarant hereby declares that all of such real property and parcels shall hereafter be held, sold, conveyed, encumbered, leased, rented, occupied and improved subject to the provisions of this Declaration.

This Declaration shall not be construed to supersede any zoning or subdivision regulation now or hereafter set forth by the City of Franklin.

ARTICLE I GENERAL

- 1.1 <u>Project Area.</u> Declarant is the owner of the property described in the attached Exhibit "A" and is or may become the owner of all or portions of the property described in the attached Exhibit "B."
- 1.2 <u>Purpose of Declaration</u>. Property which becomes subject to this Declaration in the manner hereinafter provided shall be referred to as the "Master Association Area." This Declaration is executed (a) in furtherance of a common and general plan for the property described in the attached Exhibit "A" and for those other parcels of the property which may hereafter become part of the Master Association Area; (b) to protect and enhance the quality, value, desirability and attractiveness of all property which becomes part of the Master Association Area; (c) to provide for a Master Association as a master property owners' association and vehicle to hold, maintain, care for and manage Master Association Properties and to perform functions for the benefit of Owners of Sites within the Master Association Area; (d) to define the duties, powers and rights of the Master Association Area.
- 1.3 General Statement. Declarant, for itself and its heirs, legal and personal representatives, successors and assigns, hereby declares that the property described in the attached Exhibit "A" and any parcels of property which may become subject to this Declaration in the manner hereinafter provided, and each part thereof, from the date any such property becomes subject hereto shall be owned, held, transferred, conveyed, sold, leased, rented, hypothecated, encumbered, used, occupied, maintained, altered and improved subject to the covenants, conditions, restrictions, limitations, reservations, exceptions, equitable servitudes and other provisions set forth in this Declaration, for the duration thereof, all of which are hereby declared to

be part of, pursuant to and in furtherance of a common and general plan of development, improvement, enhancement and protection of the First Subdivision and any portion of the Annexable Area which may hereafter become part of the Master Association Area. The provisions of this Declaration are intended to and shall run with the Master Association Area and, until their

expiration in accordance with the terms of this Declaration, shall bind, be a charge upon and inure to the mutual benefit of: (a) all of the property which becomes part of the Master Association Area and each part or parcel thereof; (b) Declarant and its heirs, legal and personal representatives, successors and assigns; (c) the Master Association and its successors and assigns; and (d) all persons having or acquiring any right, title or interest in any property which becomes part of the Master Association, or any part or parcel thereof, or any improvement thereon, and their respective heirs, personal and legal representatives, successors and assigns. Unless otherwise expressly provided in this Declaration, words and phrases used herein shall have the meanings as defined in Article II hereof.

ARTICLE II DEFINITIONS

- 2.1 Administrative Functions. "Administrative Functions" shall mean all functions of, for and on behalf of the Master Association that are necessary or proper under this Declaration, and shall include, without limitation: (a) providing management and administration of the Master Association; (b) providing development review, control and approval services under Article X or any other portion hereof; (c) incurring reasonable attorneys' fees and accountants' fees; (d) obtaining insurance and bonds; (e) paying real estate, personal property or other taxes levied against the Master Association Properties; (f) incurring filling fees, recording costs and bookkeeping fees; (g) obtaining and maintaining offices and office furniture and equipment; and (h) performing other reasonable and ordinary administrative tasks associated with the operation of the Master Association.
- 2.2 <u>Annexable Area.</u> "Annexable Area" shall mean all of the real property described on the attached Exhibit "B", all or any portion of which may from time to time be made subject to this Declaration pursuant to the provisions of Article III hereof. The Annexable Area may also be expanded as provided in Section 3.5 hereof.
- 2.3 Annexed Property. "Annexed Property" shall mean any portion of the Annexable Area which becomes subject to this Declaration in accordance with Section 3.3 hereof.
 - 2.4 Applicant. "Applicant" shall have the meaning described in Section 10.5 hereof.
- 2.5 <u>Appointment Period</u>. "Appointment Period" shall mean the period of time commencing as of the date of the Recordation of this Declaration and continuing until the earlier of (a) the date which is the tenth (10th) anniversary of the date of Recordation of this Declaration; (b) the date upon which Declarant ceases to own more than forty acres of land then zoned any classification other than for a use that is primarily residential within the First Subdivision or Annexable Area; or (c) the date of the Recording of Declarant's relinquishment of its right to appoint members to the Development Review Committee.
- 2.6 Area. "Area" shall mean the total area of a Site, measured to the nearest one-hundredth of an acre, exclusive of any public road right-of-way, as reasonably determined by the Master Association.
- 2.7 <u>Articles of Incorporation</u>. "Articles of Incorporation" shall mean the Charter of the Master Association, which have been or will be filed in the office of the Secretary of State of the State of Tennessee, as the same may be amended from time to time.

- 2.8 <u>Assessment</u>. "Assessment" shall mean a Common Assessment, a Special Assessment or a Reimbursement Assessment.
- 2.9 <u>Assessment Year</u>. "Assessment Year" shall mean the calendar year or such other period of twelve consecutive calendar months selected by the Board for the levying, determining or assessing of the annual Assessments under this Declaration.
 - 2.10 Board. "Board" shall mean the Board of Directors of the Master Association.
- 2.11. <u>Budget</u>. "Budget" shall mean a written, reasonably itemized estimate of the expenses to be incurred by the Master Association in the performing of its functions under this Declaration and prepared pursuant to Section 8.2 hereof.
- 2.12 <u>Bylaws</u>. "Bylaws" shall mean the Bylaws of the Master Association which have been or will be adopted by the Board, as such Bylaws may be amended from time to time.
- 2.13 <u>Class A Owner</u>. "Class A Owners" shall have the meaning set forth in Section 4.4 hereof. The "Class A membership" comprises all Class A Owners.
- 2.14 <u>Class B Owner</u>. "Class B Owner" shall have the meaning set forth in Section 4.4 hereof. The "Class B membership" is the Class B Owner's interest in the Master Association.
- Common Assessment. "Common Assessment" shall mean any assessment made for the purposes of covering the portion of the annual or other periodic costs of operating the Master Association, including expenses incurred in connection with any Administrative Function and other expenses duly incurred by or on behalf of the Master Association which are to be paid by each Owner to the Master Association for the purposes provided herein and charged to such Owner and to the Site of such Owner. Expenses which may be duly incurred on behalf of the Master Association in connection with the performance of Administrative Functions include, without limitation, the following: (a) utility charges for utilities serving the Master Association Properties and for the lighting of streets throughout the Annexable Area, as well as charges for other common services for the Master Association Properties; (b) the expenses of maintenance, operation, repair and replacement of the Master Association Properties, including, without limitation, costs of labor, equipment and materials incurred in connection therewith; (c) principal, interest and other charges payable with respect to loans to the Master Association to perform any Administrative Function, including, without limitation, loans financing the construction of improvements for the Master Association Properties; (d) such other expenses as may be determined from time to time by the Board to be incurred necessarily or appropriately for the performance of Administrative Functions, including, without limitation, taxes, utility charges, and government charges not separately assessed against Sites; and (e) the establishment and maintenance of a reasonable reserve fund or funds (i) for maintenance, repair and replacement of those portions of the Master Association Properties and improvements thereon that are the responsibility of the Master Association and that must be maintained, repaired or replaced on a periodic basis; and (ii) to cover unforeseen operating contingencies or deficiencies arising from unpaid assessments or liens, as well as from emergency expenditures and other matters, all as may be authorized from time to time by the Board. Notwithstanding the preceding sentences in this Section, Common Assessments shall not be used for the payment of any expenses for the acquisition of Master Association Properties or any material capital improvements.
- 2.16 County Records. "County Records" shall mean the Recorded real estate records in the office of the Register's Office for Williamson County, Tennessee.

- 2.17 <u>Declarant.</u> "Declarant" shall mean a corporation, and its successors and assigns. A Person shall be deemed a "successor and assign" of Cool Springs Real Estate Associates, L.P. ("Cool Springs Real Estate Associates, L.P.") as Declarant only if specifically so designated in a duly Recorded written instrument as a successor or assign of Declarant under this Declaration and/or under a Supplemental Declaration and shall be deemed a successor or assign of Declarant only as to the particular rights or interests of Declarant under this Declaration or under such Supplemental Declaration which are specifically designated in such Recorded written instrument. Notwithstanding the foregoing, a successor to Cool Springs Real Estate Associates, L.P. by reason of consolidation or merger shall be deemed a successor or assign of Cool Springs Real Estate Associates, L.P. as Declarant under this Declaration.
- 2.18 <u>Declaration</u>. "Declaration" shall mean this Master Declaration of Protective Covenants for Cool Springs, as it may be amended from time to time.
- 2.19 <u>Delegate</u>. "Delegate" shall mean the natural person selected by an Owner or Owners pursuant to Section 4.5 and 4.6 hereof to represent and cast votes on behalf of such Owner or Owners as provided in this Declaration.
- 2.20 <u>Delegate District</u>. "Delegate District" shall mean a geographical area not smaller than twenty-five acres which constitutes a portion or portions of the Master Association Area and from which all Owners in that Delegate District shall elect a single Delegate to represent their Voting Units, as further provided in Article IV hereof.
- 2.21 <u>Development Review Committee</u>. "Development Review Committee" shall mean the committee for which provision is made in Article X of this Declaration.
- 2.22 <u>Development Review Committee Representative</u>. "Development Review Committee Representative" shall have the meaning set forth in Section 10.27 hereof.
- 2.23 <u>First Mortgage</u>. "First Mortgage" shall mean the unreleased Mortgage of Record encumbering a Site which has the first and superior lien priority over all other unreleased Mortgages of Record encumbering such Site.
- 2.24 <u>First Mortgagee</u>. *First Mortgagee* shall mean the Mortgagee under a First Mortgage.
- 2.25 <u>First Subdivision</u>. "First Subdivision" shall mean all of the real property described in the attached "Exhibit "A."
- 2.26 Foreclosure. "Foreclosure" shall mean, without limitation, (i) the judicial foreclosure of a Mortgage; (ii) the exercise of a power of sale contained in any Mortgage; (ii) the conveyance of the Property unencumbered by a Mortgage to the Mortgagee thereunder in lieu of foreclosure thereof; or (iv) any action commenced or taken by a lessor to regain possession or control of a Site leased to a lessee in a transaction commonly known as a "sale/leaseback."
- 2.27 <u>Improvement to Property</u>. "Improvement to Property" shall mean any Improvement, change, alteration or addition to any property within the Master Association Area, as more particularly defined in Section 10.2 hereof.
- 2.28 <u>improvements</u>. "Improvements" shall mean all structures and any appurtenances thereto of every type or kind including, but not limited to buildings, outbuildings, garages, porches, sheds, swimming pools, patios, patio covers, awnings, painting or staining of any exterior surfaces of any visible structure, additions, walkways, bicycle trails, sprinkler systems or pipes, roads, curbing, paving, driveways, parking areas, fences, screening walls, retaining walls, stairs, decks,

fixtures, landscaping hedges, windbreaks, plantings, planted trees and shrubs, poles, signs, exterior tanks, solar panels and equipment, exterior evaporative coolers, air conditioning and external water softener fixtures, wind mills, and exterior antennas, aerials and other equipment for the reception or transmission of radio, television, microwave or other similar communication systems. "Improvements" shall also include (i) any excavation, fill, ditch, diversion, dam, berm, or anything or any device that alters the natural flow of any water in any natural or artificial drainage channel from, or upon, any other Site; and (ii) any change in the grade of any Sites of more than six inches from that existing at the time of purchase by each Owner.

- 2.29 <u>Master Association</u>. "Master Association" shall mean Cool Springs East Master Association, a Tennessee nonprofit corporation, its successors and assigns.
- 2.30 <u>Master Association Area</u>. "Master Association Area" shall mean the First Subdivision, together with all other real property, If any, which hereafter becomes subject to this declaration pursuant to the provisions of Article III hereof.
- 2.31 <u>Master Association Properties</u>. "Master Association Properties" shall mean (a) all real and personal property, including improvements, now or hereafter owned by the Master Association or with respect to which the Master Association holds an easement for the use, care or maintenance thereof, held for the common use and enjoyment of all or certain of the Owners as provided herein and/or for other purposes as may be permitted by this Declaration; and (b) any median strip of land within the right-of-way of any portion of any pubic street, which portion is within the Master Association Area or upon which any Site fronts, and which shall have been accepted by the Board as a "Master Association Property."
- 2.32 Mortgage. "Mortgage" shall mean any unreleased mortgage or deed to secure debt or other similar instrument of Record, given voluntarily by the Owner of a Site, encumbering all or any portion of the Site to secure the performance of an obligation or the payment of a debt and which is required to be cancelled upon performance of the obligation or payment of the debt, or any lease of all or any portion of a Site in a transaction commonly known as a "sale/leaseback." "Mortgage" shall not include a judgment lien, mechanic's lien, tax lien or other similar lien or involuntary encumbrance upon a Site.
 - 2.33 Mortgagee. "Mortgagee" shall mean the Person who is the holder of a Mortgage.
- 2.34 <u>Mortgagor</u>. "Mortgagor" shall mean the Person who grants an interest in such Person's property to another under a Mortgage (i.e. the maker or grantor of a Mortgage).
- 2.35 <u>Notice and Hearing</u>. "Notice and Hearing" shall mean a written notice and the reasonable opportunity for a public hearing before the Board or a tribunal appointed by the Board, in the manner provided in the Bylaws.
- 2.38 Notice of Completion. "Notice of Completion" shall mean the written notice and architect's certification to the Development Review Committee concerning the completion of any improvement to Property pursuant to the provisions of Section 10.18 hereof.
- 2.37 Notice of Noncompliance. "Notice of Noncompliance" shall have the meaning set forth in Section 10.20 hereof.
- 2.38 Notice of Withdrawal. "Notice of Withdrawal" shall have the meaning set forth in Section 3.4 hereof.
- 2.39 Owner shall mean the Person, or if more than one, all Persons collectively, who hold fee simple title of Record to a Site, except that if any Site and any

improvement thereon are owned by separate Person, then the owner of the fee simple title to the building located on the Site shall be deemed the "Owner" of such Site. A Person having an interest in any Site or any improvements located thereon merely as a security for the performance of obligations shall not be deemed an "Owner" unless such Person is a Mortgagee in possession following a default under such security obligations or has acquired the fee simple title to the Site by Foreclosure. "Owner" shall also include Declarant as long as Declarant retains ownership of all or any portion of the Master Association Area.

- 2.40 <u>Person</u>. "Person" shall mean a natural person, a corporation, a partnership, trust or any other entity, or any combination thereof.
- 2.41 Record, Recorded or Recordation. "Record" or "Recorded" or "Recordation" shall interchangeably mean the filling for record of any document in the County Records.
- 2.42 Reimbursement Assessment. "Reimbursement Assessment" shall mean a charge against a particular Owner and such Owner's Site for the purpose of reimbursing the Master Association for expenditures and other costs of the Master Association for incurring one or more of Owner's violations of this Declaration, any Supplemental Declaration, the Articles of Incorporation, the Bylaws or the Rules and Regulations, pursuant to the provisions of Section 8.10 hereof, together with late charges, interest and costs as provided for herein.
- 2.43 Related User. "Related User" shall mean a Person who obtains all or certain rights of an Owner by reason of such Person validly claiming or being entitled to such rights by, through or under such Owner. Without limiting the generality of the foregoing, "Related User" shall include any occupant, tenant, family member or contract purchaser of an Owner who occupies all or part of or resides on any Site of such Owner and any natural person who is a guest or invitee of such Owner or of such Person.
- 2.44 Rules and Regulations. "Rules and Regulations" shall mean the guidelines, rules and regulations adopted by the Board, the Delegates or the Development Review Committee, from time to time, pursuant to the provisions of Section 5.14 hereof.
- 2.45 <u>Site</u>. *Site* shall mean a parcel of land within the Master Association Area as divided and subdivided pursuant to Recorded subdivision maps, as they from time to time become current, or by deeds of conveyance.
- 2.48 <u>Special Assessment</u>. "Special Assessment" shall mean a charge against the Owner and such Owner's Site representing a portion of the costs incurred by the Master Association for the purposes set forth in Section 8.9 hereof.
- 2.47 <u>Subassociation</u>, Subassociation shall mean any Tennessee profit or nonprofit corporation, or unincorporated association, and its successors or assigns, organized and established or authorized pursuant to, or in connection with, one or more Supplemental Declarations, and the membership of which comprises Owners of Sites within all or part of the area burdened by the Supplemental Declarations.
- 2.48 <u>Supplemental Common Assessment</u>. "Supplemental Common Assessment" shall have the meaning set forth in Section 8.8 hereof. Supplemental Common Assessments are a type of Common Assessments.
- 2.49 <u>Supplemental Declaration</u>. "Supplemental Declaration" shall mean a written instrument (including any amendments thereto) containing covenants, conditions, restrictions, reservations, easements, equitable servitudes or other provisions, or any combination thereof, which complies with the requirements of Section 3.3 hereof and which is Recorded either with

respect to the First Subdivision or in order to submit any portion of the Annexable Area to the provisions of this Declaration, thereby causing such portion to become part of the Master Association area.

- 2.50 <u>Utility Easement Area.</u> "Utility Easement Area" shall mean a strip of land ten feet in width running inside and along all of the boundary lines of each Site, except any Site fronting Carothers Parkway or Cool Springs Boulevard.
- 2.51 <u>Voting Unit.</u> "Voting Unit" shall mean a unit of the voting rights of a Class A Owner appurtenant to its Site so that, in accordance with Sections 4.4 and 4.6 hereof, each acre (rounded to the nearest whole number of acres) of each Site that is then zoned for a use that is primarily residential owned by one or more Class A Owners shall be deemed to have one Voting Unit and each acre (rounded to the nearest whole number of acres) of each Site that is then zoned any classification other than for a use that is primarily residential owned by one or more Class A Owners shall be deemed to have two Voting Units; provided that to the extent a Class A Owner owns less than one hundred percent of the fee simple title to any portion of a Site, such Owner's Voting Units appertaining to such site shall be determined by multiplying the acreage jointly owned by such Owner by such Owner's undivided interest in and to such acreage and then applying the standards set forth above in this Section. Voting Units shall not be fractionalized.

ARTICLE III ANNEXATION TO MASTER ASSOCIATION AREA

- 3.1 <u>First Subdivision Made Subject to Declaration</u>. Declarant hereby declares that the First Subdivision is hereby made subject to this Declaration.
- 3.2 Property Which May Be Annexed. From time to time Declarant may unilaterally add to the Master Association Area and make subject to this Declaration all or any portion or portions of the Annexable Area. The portion of the Annexable Area covered by a particular Supplemental Declaration need not adjoin the First Subdivision or any other portion of the Annexable Area covered by a Supplemental Declaration Recorded prior to the particular Supplemental Declaration. Prior to the commencement of the construction of any improvements on any portion of the Annexable Area after the sale of such portion to a Person who is not a "successor or assign" of Cool Springs Real Estate Associates, L.P. as Declarant as defined in Section 2.17 hereof, Declarant shall arrange to have such portion added to the Master Association Area if Declarant shall determine that the use to be made of such portion is compatible with the uses of the Master Association Area permitted by this Declaration and that such portion is sufficiently physically contiguous to the Master Association Area to enable such portion to be developed as an integral part of the development planned for Cool Springs Real Estate Associates, L.P.
- 3.3 Manner of Annexation. Any parcel of real property (the "Annexed Property") within the Annexable Area may, from time to time, become part of the Master Association Area and become subject to this Declaration effective upon the Recordation of a Supplemental Declaration meeting the requirements hereinafter set forth. Each Supplemental declaration shall: (a) be executed by the then Owner or Owners of the Annexed Property described therein; (b) during the Appointment Period, also contain the executed written consent of Declarant if the Annexed Property is not then owned by Declarant, unless such written consent is waived by Declarant with respect to all or any portion of the Annexable Area by a writing signed by Declarant and Recorded in the County Records, and after the expiration of the Appointment Period, also contain the written consent of the Master Association; (c) contain an adequate legal description of the Annexed Property; (d) contain a reference to this Declaration which shall state its date of

Recordation and the book and page of the County Records where this Declaration is Recorded; (e) contain a statement that the Annexed Property is declared to be part of the Master Association Area under this Declaration and that the Annexed Property shall be subject to this Declaration; (f) designate the Delegate District in which the Annexed Property is located; (g) provide that Sites therein shall be subject to the jurisdiction of a Subassociation, or shall not be subject to the jurisdiction of a Subassociation, as the case may be; and (h) provide for the method of amendment, which shall require the written consent of Declarant during the Appointment Period, and thereafter shall require the written consent of the Master Association. A Supplemental Declaration may provide for phased annexation so that separate parcels of real property may be made subject to such Supplemental Declaration and this Declaration at different times. A Supplemental Declaration may impose on the Annexed Property described therein additional covenants, conditions, restrictions, limitations, reservations, exceptions, equitable servitudes and provisions other than those set forth in this Declaration, taking into account the unique and particular aspects of the proposed development of the Annexed Property covered thereby. A Supplemental Declaration may provide for a Subassociation of Owners within the Annexed Property and for the right of the Subassociation to assess such Owners and to place liens upon the Sites of such Owners. Upon Recordation of a Supplemental Declaration, the Annexed Property shall be subject to all of the covenants, conditions, restrictions, limitations, reservations. exceptions, equitable servitudes and other provisions set forth in this Declaration, except to the extent, if any, specifically stated in the Supplemental Declaration.

- Withdrawal of Annexed Property by Declarant. Annexed Property for which a 3.4 Supplemental Declaration has been Recorded may be withdrawn from the Master Association Area, from this Declaration and/or from such Supplemental Declaration by Declarant to correct a survey or platting error or other technical or clerical errors. The withdrawal of such Annexed Property may be accomplished by the execution and Recordation of a written notice of such withdrawal (the "Notice of Withdrawal"); provided that (a) no vote has then been exercised with respect to Owners (excluding Declarant) of Sites located within such Annexed Property and (b) no Assessments to the Master Association have then commenced with respect to such Owners (excluding Declarant) of Sites located within such Annexed Property. Any such Notice of Withdrawal shall (i) be executed and acknowledged by the Owner or Owners (including Declarant) of such Annexed Property; (ii) if such Annexed Property is not then owned by Declarant, contain the executed and acknowledged written consent of Declarant during the Appointment Period; (iii) contain an adequate legal description of such Annexed Property; (iv) contain the specific reason for such withdrawal; (v) contain a reference to the Supplemental Declaration for such Annexed Property, which reference shall state the date of Recordation thereof and the book and page of the County Records where the Supplemental Declaration was Recorded; and (vi) contain a statement and declaration that such Annexed Property is thereby withdrawn from the Master Association Area and shall not be thereafter subject to this Declaration or the Supplemental Declaration for such Annexed Property. The withdrawal shall be effective upon the Recording of the Notice of Withdrawal, and upon such Recording, such Annexed Property described therein shall no longer be part of the Master Association Area or subject to this Declaration or to the Supplemental Declaration for such Annexed Property, as more fully set forth in such Notice of Withdrawal.
- 3.5 Expansion of Annexable Area. The Annexable Area may be expanded to add additional real property which is either contiguous to or separated only by a road right-of-way or body of water from the Annexable Area as it then is configured. Any such expansion shall be effective upon the Recordation of a written instrument executed by Declarant during the Appointment Period and all other owners of fee simple title to such additional real property, describing such additional real property and declaring that such additional real property shall thereafter be added to the Annexable Area.

ARTICLE IV MASTER ASSOCIATION OPERATIONS

- 4.1 <u>Master Association</u>. The Master Association has been or will be formed as a Tennessee corporation under the Tennessee Non-profit Corporation Act. The Master Association shall have the duties, powers and rights set forth in this Declaration and in the Articles of Incorporation and Bylaws.
- 4.2 Board. The affairs of the Master Association shall be managed by the Board, which shall consist of neither fewer than three nor more than five Directors. Notwithstanding any provision of this Declaration to the contrary, during the Appointment Period, the number of Directors shall be three and Declarant shall have the right to appoint all three of such Directors. Subject to the foregoing, the number, term, election and qualifications of the Board shall be fixed in the Articles of Incorporation or the Bylaws or both. By resolution, the Board may delegate portions of its authority to an executive committee or to other committees, tribunals, officers of the Master Association or to agents and employees of the Master Association, but such delegation of authority shall not relieve the Board of the ultimate responsibility for management of the affairs of the Master Association. Action by or on behalf of the Master Association may be taken by the Board or any duly authorized executive committee, officer, agent or employee without a vote of Delegates, except as otherwise specifically provided in this Declaration.
- 4.3 <u>Membership in Master Association</u>. Each Owner of a Site within the Master Association Area shall be a member of the Master Association. There shall be one membership in the Master Association for each Site within the Master Association Area. The Person or Persons who constitute the Owner of a Site shall automatically be the holder of the membership in the Master Association appurtenant to that Site, and such membership shall automatically pass with during the Appointment fee simple title to the Site. Membership in the Master Association shall not be assignable separate and apart from fee simple title to a Site, except that an Owner may assign some or all of such Owner's right as an Owner to use improvements or otherwise to a Related User or Mortgagee and may arrange for a Related User to perform some or all of such Owner's obligations as provided in this Declaration, but no Owner shall be permitted to relieve itself of the responsibility for fulfillment of all of the obligations of an Owner under this Declaration.
- 4.4 Classes of Membership. The Association shall have two classes of voting membership, Class A Owners and Class B Owners. Class A Owners shall include all Owners; provided, however, that so long as Declarant is a Class B Owner, it shall not be entitled to Class A membership, unless Declarant owns a Site on which a building is being or has been constructed, and then Declarant shall be entitled to Class A membership only as to such Site. Class A Owners shall be entitled to vote their respective Voting Units. The Class B Owner shall be the Declarant, and the Class B membership shall exist until the earlier of (a) the date which is the tenth (10th) anniversary of the date of Recordation of this Declaration; (b) the date upon which Declarant ceases to own more than 40 acres of land then zoned any classification other than for a use that is primarily residential within the First Subdivision or Annexable Area; or (c) the date of the Recording of Declarant's relinquishment of its right for its Delegate to have one more vote than the aggregate number of votes which all other Delegates are entitled to cast, as provided in Section 4.7 hereof. Notwithstanding anything herein stated to the contrary, at such time as Class B membership ceases to exist, Declarant shall become a Class A Owner with respect to the property in the Master Association Area it then owns, if any.
- 4.5 Establishment of Delegate Districts. The Master Association Area shall be divided into Delegate Districts, and each Delegate District shall elect one Delegate to the Master Association to exercise the voting power of all the Owners in such Delegate District. Unless otherwise provided in the Supplemental Declaration by which a Subassociation is created, all of the Annexed Property within the jurisdiction of such Subassociation shall constitute a Delegate

District. A Delegate District may consist of one or more Annexed Properties. A Subassociation may have jurisdiction over one or more Annexed Properties and/or one or more Delegate Districts. In the event that a Subassociation shall not be created for any portion of the Annexed Property, then the Delegate District or Districts for such Annexed Property shall be established by Declarant by the Recordation of one or more Supplemental Declarations or other written Instruments signed by Declarant. Such Supplemental Declarations or other instruments shall contain legal descriptions of the portions of the Annexed Property which shall be or become part of a Delegate District and a statement that such real property described therein shall be or become part of a designated Delegate District for purposes of this Declaration.

- Units for the election of the Delegate to the Master Association to exercise the voting power of the Delegate District in which the Owner's Site is located. If such Delegate District is within the jurisdiction of a Subassociation, then the Class A Owner shall have: (a) the right to vote for the election of the Delegate from that Delegate District and the right to vote on Delegate District matters as set forth in this Declaration and (b) the Class A Owner shall have the right to vote for the election of the board of directors of the Subassociation and the right to vote on Subassociation matters as set forth in the Supplemental Declaration for such Subassociation. The Class B Owner shall be entitled to designate one Delegate so long as the Class B Membership exists. The election of the Delegate to represent any Delegate District and the voting on Delegate District matters shall be made by Class A Owners holding a majority of the Voting Units in such Delegate District. The Bylaws shall provide for the manner, time, place, conduct and voting procedures for meetings of Class A Owners for the purpose of electing a Delegate in any such Delegate District.
- Voting Rights of Delegates. Each Delegate shall be entitled to cast one vote for each Voting Unit appurtenant to each Site which is owned by a Class A Owner and which is subject to this Declaration and located in the Delegate District represented by such Delegate, except that so long as the Class B membership exists, the Delegate designated by the Class B Owner shall be entitled to one more than the aggregate number of votes which all other Delegates are then entitled to cast. At each meeting of Delegates, each Delegate shall cast the votes which he or she is entitled to cast in such manner as such Delegate may, in his or her sole and reasonable discretion, deem appropriate, acting on behalf of all the Owner(s) such Delegate represents; however, in the event that any Class A Owner in attendance in person or by proxy at any duly constituted meeting of the Class A Owners in a Delegate District shall so request with respect to the vote on any issue to be considered by the Delegates, the Delegate representing such Delegate District shall cast all of the votes of such Delegate District in the same proportion as the Class A Owners in such Delegate District shall have, in person or by proxy, cast their Voting Units in favor of or in opposition to such issue. A Delegate shall have the authority, in his or her sole discretion, to call a special meeting of the Class A Owners in such Delegate's Delegate District in the manner provided in the Bylaws for the purpose of obtaining instructions as to the manner in which such Delegate is to vote on any issue to be voted on by the Delegates. When a Delegate is voting in his or her own discretion, without instruction from the Class A Owners whom such Delegate represents, then such Delegate may cast all of the votes which he or she represents as a unit or such Delegate may apportion some of such votes in favor of a given proposition and some of such votes in opposition to such proposition. It shall be conclusively presumed for all purposes of Master Association business that any Delegate casting votes on behalf of the Class A Owners in such Delegate's Delegate District will have acted with the authority and consent of all such Class A Owners. All agreements and determinations lawfully made by the Master Association in accordance with the voting procedures established herein and in the Bylaws shall be binding on all Owners and their successors and assigns.

ARTICLE V DUTIES AND POWERS OF MASTER ASSOCIATION

- 5.1 General Duties and Powers of Master Association. The Master Association has been formed to further the common interests of the Owners. The Master Association, acting through the Board or through persons to whom the Board has delegated any authorized powers of the Board, shall have the duties and powers hereinafter set forth and, in general, the powers to perform its duties described in this Declaration and, subject to any limitation set forth in this Declaration, the powers to do anything that may be necessary or desirable to further the common interests of the Owners, to maintain, improve and enhance Master Association Properties and to improve and enhance the attractiveness, desirability and safety of the Master Association Area.
- Association shall accept title to any property, including any improvements thereon and personal property, transferred to the Master Association by Declarant, together with the responsibility to perform any and all Administrative Functions associated therewith, provided that such property and Functions are not inconsistent with the provisions contained in this Declaration. Property interests transferred to the Master Association by Declarant may include fee simple title, easements, leasehold interests and licenses to use. Any property or interest in property transferred to the Master Association by Declarant shall be unencumbered by any Mortgage and shall not impose any unreasonable or special burden on the Master Association other than the normal burdens of ownership of property.
- operate, care for, maintain and repair all Master Association Properties and keep them in a safe, attractive and desirable condition for the use and enjoyment of the Owners. The Master Association shall have a reasonable right of entry upon any Site to make emergency repairs and to do other work reasonably necessary under this Declaration or under any applicable Supplemental Declaration for the proper performance of its duties hereunder and the proper maintenance and operation of the Master Association Properties.
- 5.4 <u>Duty to Pay Taxes</u>. The Master Association shall pay all ad valorem taxes and governmental assessments levied upon the Master Association Properties to which the Master Association holds fee simple title and all taxes and assessments payable by the Master Association. Nevertheless, the Master Association shall have the right to contest in good faith any such taxes or assessments.
- 5.5 <u>Casualty Insurance</u>. To the extent deemed desirable by the Board, the Master Association shall obtain and keep in full force and effect at all times, to the extent reasonably obtainable, casualty, fire and extended coverage insurance with respect to all insurable improvements and personal property owned by the Master Association.
- 5.6 <u>Liability Insurance</u>. To the extent deemed desirable by the Board, the Master Association shall obtain and keep in full force and effect at all times, to the extent reasonably obtainable, broad form comprehensive liability insurance covering public liability for bodily injury and property damage.
- 5.7 General Provisions Respecting Insurance. Insurance obtained by the Master Association may contain such deductible provisions as good business practice may dictate. Insurance obtained by the Master Association shall, to the extent reasonably possible without undue cost, contain a waiver of rights of subrogation as against the Master Association and each Owner as against any officer, director, agent or employee of any of the foregoing. To the extent reasonably possible, and provided Declarant reimburses the Master Association for any additional premium payable on account thereof, insurance obtained by the Master Association shall name

Sidewalks for primarily pedestrian use shall have an unobstructed width of at least five feet and shall be provided as per city standard. All sidewalks shall meet accessibility standards for disabled users as required by appropriate government authorities. When multiple uses are anticipated on the same walk / trail, the width shall be increased to minimize conflicts between the various users. Twelve feet widths are recommended in such instances, with eight feet, minimum, to accommodate multiple uses.

1. Special Requirements for Pedestrian Circulation:

- a. "Village Center / Town Square": special paving materials and patterns should be utilized; increase width to minimum width of twelve feet with a four foot zone, at back of curb, for street furnishings (signs, street trees, trash receptacle, light standards, etc.)
- b. Single family and Multifamily: internal walks and connectors to other areas are required throughout any single family sub-division on both sides of all streets. Multifamily development shall incorporate pedestrian walkways throughout, including pedestrian circulation in public right-of-ways adjacent to each property.
- c. Commercial: all commercial areas shall provide adequate pedestrian circulation within the parcel development, as well as the public right-ofway pedestrian circulation network adjacent to each property.

I. Building Orientation

As part of an integrated approach to site planning, the orientation of any building(s) on any site must be considered crucial to the success of the development as well as its impact on the site. Views to and from roadways, surrounding parcels and significant existing or proposed site features (vegetation, other amenities) should be incorporated into the siting of buildings to limit any potential negative impacts while taking advantage of any benefits.

Site and building entries should be considered as a sequential experience, which is logical and easily understood by all. As appropriate, the separation of various functions can add clarity to how efficiently a site works as well as its aesthetic impression. This type of approach may include separation of service, visitor and employee entries depending on the specific nature of the sites use.

Service areas, loading docks and storage areas shall be incorporated into the building design and oriented so that they are screened from adjacent right-of-ways and properties.

Rules and Regulations and the provisions of this Declaration, the provisions of this Declaration shall prevail.

- Power to Enforce Declaration, Supplemental Declaration and Rules and 5.15 Regulations. The Master Association shall have the power to enforce the provisions of this Declaration and the provisions of the Rules and Regulations and shall take such action as the Board deems necessary or desirable to cause such compliance by each Owner and each Related User. Without limiting the generality of the foregoing, the Master Association shall have the power to enforce the provisions of this Declaration and of the Rules and Regulations by any one or more of the following means: (a) by commencing and maintaining actions and suits to restrain and enjoin any breach or threatened breach of the provisions of this Declaration or the Rules and Regulations, by mandatory injunction or otherwise; (b) by commencing and maintaining actions and suits to recover damages for breach of any of the provisions of this Declaration or the Rules and Regulations; (c) by levying and collecting reasonable and uniformly applied fines and penalties, established in advance in the Rules and Regulations, from any Owner or Related User for breach of this Declaration or the Rules and Regulations by such Owner or such Related User. and (d) by exercising any remedy or remedies for nonpayment of Assessments pursuant to Section 8.12 hereof.
- 5.18 Power to Provide Services to Subassociations. The Master Association shall have the power to provide services to Subassociations. Such services to any Subassociation shall be provided pursuant to the request of a Subassociation or the Owners governed by such Subassociation, as evidenced by a Supplemental Declaration, or pursuant to an agreement in writing between the Master Association and such Subassociation. Any arrangement for the rendition of such services shall provide for the payment by such Subassociation to the Master Association of the reasonably estimated expenses of the Master Association for providing such services to the Subassociation, including a fair share of the overhead expenses of the Master Association, and a reasonable amount of profit for the Master Association. Services which may be provided to the Subassociation may include without limitation: (a) the construction, care, operation, management, maintenance, repair and replacement of Improvements owned by the Subassociation and the maintenance, management and care of real property in any area governed by the Subassociation; (b) the enforcement of the provisions of any Supplemental Declaration for, on behalf of and in the name of the Subassociation; (c) the collection of assessments for, in the name of and on behalf of a Subassociation; (d) the payment of taxes for a Subassociation with funds of the Subassociation; (e) the obtaining of insurance for a Subassociation; (f) the collection of charges for use of facilities of a Subassociation; and (g) the appointment and supervision of a manager or managers for a Subassociation.
- 5.17 Power to Provide Special Services for Owners. The Master Association shall have the power to provide services to an Owner or a group of Owners. Any service or services to an Owner or group of Owners shall be provided pursuant to a Supplemental Declaration or pursuant to an agreement in writing between the Master Association and such Owner or group of Owners. Any arrangement for the rendition of such services shall provide for payment to the Master Association by such owner or group of Owners of the reasonably estimated expenses of the Master Association for providing such services, including a fair share of the overhead expenses of the Master Association, plus a reasonable amount of profit for the Master Association, and shall contain reasonable provisions assuring that the obligation to pay for such services shall be binding upon any heirs, personal representatives, successors and assigns of the Owner or group of Owners and that the payment for such services shall be secured by a lien on the property of the Owner or group of Owners.
- 5.18 <u>Power to Grant Easements</u>. The Master Association shall have the power to grant permits and Ilcenses, as well as easements for access, utilities, drainage, water facilities and other matters, in, on, over, across or under Master Association Properties and the Utility Easement

Area, as may be reasonably necessary or useful for the proper maintenance of the Master Association Properties.

- 5.19 Power to Convey and Dedicate Property to Government Agencies. The Master Association shall have the power to grant, convey, dedicate or transfer any Master Association Properties or facilities to any public or governmental agency or authority for public use; however, if the means of ingress to and egress from a Site is through any such Master Association Property, then any such grant, conveyance, dedication or transfer shall be effected so as to provide ingress and egress for the benefit of the Owner of such Site.
- 5.20 <u>Power to Borrow Money</u>. The Master Association shall have the power to borrow money but not the power to encumber Master Association Properties as security for such borrowing.
- 5.21 Power to Employ Managers: Management Contracts. The Master Association shall have the power to retain and pay for the services of a manager or managers to undertake any of the management duties and Administrative Functions for which the Master Association has responsibility, and the Master Association may delegate any of its duties, powers or functions to any such manager. Notwithstanding any delegation to a manager of any duties, powers or functions of the Master Association, the Master Association and the Board shall remain ultimately responsible for the performance and exercise of such duties, powers and functions.
- 5.22 Power to Engage Employees, Agents and Consultants. The Master Association shall have the power to hire and discharge employees and agents and to retain and pay for legal, accounting and other services as may be necessary or desirable in connection with the performance of any duties or the exercise of any powers of the Master Association under this Declaration or any Supplemental Declaration.
- 5.23 General Corporate Powers. The Master Association shall have all of the ordinary powers and rights of a Tennessee nonprofit corporation, including without limitation the power and right to enter into partnerships and other agreements, subject only to such limitations upon such powers as may be set forth in this Declaration or in the Articles of Incorporation or Bylaws. The Master Association shall also have the power to do any and all lawful things which may be authorized, required or permitted to be done under this Declaration, under any Supplemental Declaration, or under the Articles of Incorporation, Bylaws or Rules and Regulations, and to do and perform any and all acts which may be necessary or desirable for or incidental to, the exercise of any of the express powers or rights of the Master Association under this Declaration, under any Supplemental Declaration or under the Articles of Incorporation, Bylaws or Rules and Regulations.

ARTICLE VI MASTER ASSOCIATION PROPERTIES

- 6.1 Owner's Rights of Use and Enjoyment Generally. Every Owner shall have a right and easement of enjoyment in and to the Master Association Properties, which shall be appurtenant to and shall pass with the title to each Site, subject to the provisions contained in this Declaration, in any Supplemental Declaration, the Articles of Incorporation, Bylaws and the Rules and Regulations. All Owners may use the Master Association Properties, unless otherwise provided in this Declaration or unless provided in the Supplemental Declaration governing the Site of any such Owner or in the Supplemental Declaration governing a particular Master Association Property, or both.
- 6.2 Right of Master Association to Regulate Use. The Master Association, acting through the board, shall have the power to regulate the use of Master Association Properties by

Owners or Related Users of Owners to further and enhance the overall rights of use and enjoyment of all Owners, including imposing reasonable limits on the times of use and number of Persons permitted to use Master Association Properties.

- 6.3 Right of Master Association to Allow Public Use. Subject to the provisions of Section 5.18 hereof, the Master Association, acting through the Board, shall have the right to allow members of the general public to use Master Association Properties, subject to reasonable limitations, and provided that use by the general public does not unreasonably interfere with or impair the rights of use and enjoyment of Owners.
- 6.4 No Partition of Master Association Properties. No Owner shall have the right to partition or seek partition of the Master Association Properties or any part thereof.
- Master Association for any damage to Master Association Properties or for any expense or liability incurred by the Master Association, to the extent not covered by insurance, which may be sustained by reason of the negligence or willful misconduct of such Owner and for any violation by such Owner of this Declaration, any Supplemental Declaration, the Articles of Incorporation, the Bylaws or the Rules and Regulations. The Master Association shall have the power, as elsewhere provided in this Declaration to levy and collect a Reimbursement Assessment against an Owner to cover the costs and expenses incurred by the Master Association on account of any such damage or any such violation of this Declaration, Bylaws or Rules and Regulations, or for any increased insurance premiums directly attributable to any such damage or any such violation.
- Damage, Destruction or Required Improvements to Master Association Properties. 8.8 In the event of damage to Master Association Properties by fire or other casualty or in the event any governmental authority shall require any repair, reconstruction or replacement of any Master Association Properties, the Master Association shall have the duty to repair, reconstruct or replace the same. Subject to the provisions of Section 5.7 hereof, any Insurance proceeds payable by reason of damage or destruction of Master Association Properties by fire or other casualty shall be paid to the Master Association and shall be used, to the extent necessary, to pay the costs of repair, reconstruction or replacement. If funds from Insurance proceeds or from available reserves are insufficient to pay all costs of repair, reconstruction or replacement of improvements damaged or destroyed, or if the Master Association is required to make repairs, replacements or improvements by governmental authorities, the Master Association may, in order to make up and deficiency in the insurance proceeds or to pay for the required repair, replacement or Improvement, levy a Special Assessment in accordance with Section 8.09 hereof, or if an Owner or group of Owners is liable or responsible for such damage, levy a Relmbursement Assessment against the Owner or group of Owners responsible thereof, to provide the additional funds necessary as elsewhere provided in this Declaration. Repair, reconstruction or replacement of Master Association Properties shall be done under such contracting and bidding procedures as the Board shall reasonably determine are appropriate. If insurance proceeds available to the Master Association on account of damage or destruction exceed the cost of repair, reconstruction and replacement, the Master Association shall use the same for future maintenance, repair, improvement and operation of Master Association Properties.
 - Association Properties or interests therein are taken under exercise of the power of eminent domain or by private purchase in lieu thereof, the award in condemnation or the price payable shall be paid to the Master Association, except to the extent payable to any other Person with an interest in such property. The Master Association shall have the exclusive right to participate in such condemnation proceedings as they pertain to Master Association Properties and to represent the interests of all Owners in such proceedings; each Owner hereby irrevocably appoints the Master Association and any such duly appointed trustee as such Owner's attorney-in-fact for such

- purposes. All condemnation compensation, damages or other proceeds received by the Master Association shall be held by the Master Association as determined by the Board, as a reserve for future maintenance, repair, reconstruction or replacement of Master Association Properties, or may be used for improvements or additions to, or operation of, Master Association Properties; provided, however, if an allocation of such condemnation compensation, damages or other proceeds is already established in negotiation, judicial decree or otherwise, then in allocating such condemnation compensation, damages or other proceeds the Master Association shall employ such allocation.
- 6.8 <u>Title to Master Association Properties on Dissolution of Master Association.</u> In the event of the dissolution of the Master Association, the Master Association Properties shall, to the extent reasonably possible, be conveyed or transferred to an appropriate public or governmental agency or agencies, or to a nonprofit corporation, association, trust or other organization, to be used, in any such event, for the common benefit of Owners for purposes similar to those for which the particular Master Association Property was held by the Master Association. To the extent the foregoing is not possible, the Master Association Properties, and the proceeds from the sale or disposition shall be distributed to Owners in proportion to their then prevailing shares for the payment of Common Assessments.

ARTICLE VII DECLARANT'S RIGHTS AND RESERVATIONS

- Period of Declarant's Rights and Reservations. Declarant shall have, and hereby retains and reserves, certain rights as set forth in this Declaration with respect to the Master Association and Master Association Properties. The right and reservation of Declarant set forth in this Declaration shall be deemed excepted and reserved in each Recorded Supplemental Declaration, in each conveyance of property by Declarant to the Master Association and in each deed or other instrument by which any property within the Master Association Area is conveyed, whether or not specifically stated therein. The rights, reservations and easements of Declarant set forth in this Declaration shall be prior and superior to any other provisions of this Declaration and during the Appointment Period may not, without Declarant's prior written consent, be modified, amended, rescinded or affected by any amendment of this Declaration, including any amendment of this Section. Declarant's consent to any such amendment shall not be construed as consent to any other or subsequent amendment.
- Declarant shall have and hereby reserves the right, at any time and from time to time, but shall not be obligated to, construct, at its expense, additional improvements on Master Association Properties which are for the improvement and enhancement thereof and for the benefit of the Master Association and Owners, and which are not detrimental to any Owner, so long as the anticipated cost of the maintenance of any such improvements, as reasonably estimated by Declarant, would not materially increase the amount of expenses of the Master Association otherwise payable by funds raised through Common Assessments. If the means of ingress and egress from a Site is through such Master Association Property, then any such construction by Declarant shall be made subject to an easement of ingress and egress for the benefit of the Owners of such Site. Declarant shall convey or transfer such improvements to the Master Association and the Master Association shall be obligated to accept title to, care for and maintain the same as elsewhere provided in this Declaration.
- 7.3 Declarant's Rights to Use Master Association Properties in Promotion and Marketing of Master Association Area. Declarant shall have and hereby reserves the right to the reasonable use of Master Association Properties in connection with the development, construction, promotion, marketing, sale and leasing of properties within the Master Association Area, by erecting and maintaining on any part of the Master Association Properties such signs as Declarant

may reasonably deem necessary or proper in connection with the development, construction, promotion, marketing, sale and leasing of parcels of real property within the Master Association Area, and by permitting prospective purchasers of any of such parcels who are not Owners to enter upon Master Association Properties; provided, however, that Declarant shall pay all costs occasioned by such use, including without limitation maintenance and repair expenses.

- Declarant's Rights to Complete Development of Cool Springs. No provision of this Declaration shall be construed to limit the right of Declarant to or require Declarant to obtain approval to: (a) excavate, cut, fill or grade any property owned by Declarant or to construct, after, remodel, demolish or replace any improvements on any Master Association Property; provided, however, that any such activities on any Master Association Property shall not result in materially increased Master Association expenses, as reasonably estimated by Declarant, and that following any such activities on any Master Association Property, Declarant shall restore such Master Association Property to a condition no worse than the condition existing before the occurrence of such activities; or (b) require Declarant to seek or obtain the approval of the Development Review Committee or of the Master Association for any such activity or improvement to Property by Declarant on any Master Association Property. Nothing in this Section shall limit or impair the reserved rights of Declarant as elsewhere provided in this Declaration.
 - 7.5 <u>Declarant's Rights to Grant and Create Easements</u>. Declarant shall have and hereby reserves the right to grant or create temporary or permanent easements and rights-of-way for access, utilities, drainage, water and other purposes incident to development, construction or sale within the Annexable Area, located in, on, under, over and across: (a) Sites owned by Declarant and (b) Master Association Properties, provided that such easements and rights-of-way do not unreasonably interfere with the rights of Owners.
 - 7.8 Declarant's Rights to Convey Additional Property to Master Association. Declarant shall have and hereby reserves the right, but shall not be obligated to, convey additional real property and improvements thereon to the Master Association at any time and from time to time in accordance with this Declaration, so long as any such improvements are completed at the time of the conveyance of the real property on which they are located. Declarant hereby states its intent that the only properties to be conveyed by it to the Master Association will be (a) landscaped areas at public road intersections and other locations which the beautification thereof would be of general benefit to Owners, (b) areas on which are situated sign(s) erected by Declarant for the benefit of more than one Owner, and (c) takes and other retention areas.

ARTICLE VIII ASSESSMENTS

- 8.1 Covenant to Pay. Each Owner, by acceptance of a deed to his Site, whether or not it shall be so expressed in such deed, is thereby deemed to covenant and agree to pay to the Association: (a) Common Assessments, (b) Special Assessments, (c) Reimbursement Assessments, and (d) fines which may be imposed against such Site in accordance with Section 11.7 hereof.
- 8.2 Annual Budgets and Financial Reports. At least sixty days prior to the commencement of each Assessment Year, the Board shall cause a proposed Budget for such Assessment Year to be prepared. The proposed Budget shall show, in reasonable detail, the categories or expenses and the anticipated amounts of expenses for which Common Assessments are determined by the Board to be necessary or desirable, and shall reflect any expected income and estimated sources and amounts thereof of the Master Association for such Assessment Year and any expected surplus from the prior Assessment Year. The proposed Budget shall not include any expenditures for material capital improvements, new construction or new landscaping. Such proposed Budget shall include a companson of such anticipated expenses and income for the

upcoming Assessment Year with those for the then-current Assessment Year (based, to the extent possible, upon the actual income and expenses for the then-current Assessment Year). Board shall cause a copy of such proposed Budget to be sent by first-class mail to each Owner promptly after such proposed Budget is prepared and shall cause a copy thereof to be posted at the principal office of the Master Association Thereafter, each Owner may provide comments and suggestions about the proposed Budget to the Board. The Board shall adopt a Budget for the upcoming Assessment Year not less than thirty days before the beginning of such Assessment Year and promptly after such adoption shall cause a copy thereof to be sent by first-class mail to each Owner and a copy thereof to be posted at the principal office of the Master Association. In the event any Owner has not furnished its address to the Master Association, such posting shall be deemed to constitute delivery to any such Owner. At the office of the Master Association, copies of the proposed budget and the Budget shall be made available by the Master Association to any Person requesting a copy thereof upon payment of the reasonable expense of copying the same, and the books and records of the Master Association may be examined by any Owner or any authorized representative of an Owner during normal business hours. Furthermore, the Master Association shall send by first-class mall an annual cash flow statement and an annual financial statement, as certified by an officer of the Master Association, to each Owner within four months after the end of each Assessment Year.

- 8.3 <u>Levying of Common Assessments</u>. For each assessment Year, the Master Association shall, pursuant to the provisions of this Declaration and any applicable Supplemental Declaration, levy Common Assessments against all Owners.
- Assessment Year which is payable by an Owner for the Site of such Owner shall be computed by multiplying the total amount to be raised by Common Assessments for that Assessment Year, as shown in the Budget for that Assessment Year, by a fraction, the numerator of which is the Area of that Site as of the first day of that Assessment Year, and the denominator of which is the Master Association Area (less dedicated public rights of way) as of the first day of that Assessment Year, provided, however, that only one-half of the Area of Sites zoned for residential use shall be used in the determination of both such numerator and denominator. Declarant shall be responsible for paying any Common Assessments for that portion of the Master Association area owned by Declarant in accordance with the fractional percentage set forth herein above.
- Association Area, the Assessments shall commence on the first day of the first month following the date of Recordation of the first deed conveying a Site within such Assessment Area from Declarant to the first non-Declarant Owner. The Assessments for the then-current Assessment Year for each Site within such Assessment Area shall be prorated on the basis of the number of months in such Assessment Year remaining from the date of commencement of such Assessments to the end of such Assessment Year. To facilitate the determination of the Area of a Site by the Master Association, each Owner shall, upon the request of the Master Association, deliver to the Master Association a print of the most accurate and recent boundary survey of such Owner's Site available to such Owner, which survey shall show the Area of such Site to the nearest one-hundredth of an acre or less.
- payable in advance to the Master Association by the assessed Owners during the Assessment Year in annual installments, and, except for the initial Common Assessment payable for a Site, shall be due and payable on or before the first day of the first month of such Assessment Year, or in such other manner and on such other date or dates as the Board may designate in its sole and absolute discretion. Notice of the amount of the Common Assessment for a particular Assessment

Year shall be given to the Owners in a reasonable period of time prior to the date in such Assessment Year when the Common Assessment, or the first installment thereof, is due.

- 8.7 Fallure to Establish Common Assessments. The fallure by the Board to levy Common Assessments for any Assessment Year shall not be deemed a waiver or modification of any of the provisions of this Declaration or a release of the liability of any Owner to pay Common Assessments, or any installment, thereof, for that or any subsequent Assessment Year. In the event of such failure, the Budget and the amount of the Common Assessment for that Assessment Year payable for each Site shall be, until subsequently modified by the Board, the same as for the most recent year for which Common Assessments shall have been levied.
- Supplemental Common Assessments. In any Assessment Year, if the Board determines that the important and essential functions of the Master Association may not be fully funded by the Common Assessments received or receivable pursuant to the Budget for that Assessment Year, the Board may levy one or more Supplemental Common Assessments, applicable to that year only, by notifying each Owner by first-class mail of the amount of the Budget deficit, the reason(s) therefor, and the date(s) and amount(s) of such Supplemental Common Assessment payable by such Owner. The due date for payment of any such Supplemental Common Assessment shall be as specified in the resolution authorizing such assessment, but not earlier than thirty days after the sending of notice to the Owners in accordance with the preceding sentence; provided, however, that the Board may make Supplemental Common Assessments payable in installments over a period that may, in the Board's discretion, extend beyond the Assessment Year in which adopted. The proportion of each Supplemental Common assessment to be paid by an Owner of a Site shall be equal to the percentage of the annual Common Assessment for such Site for the Assessment Year in which the Supplemental Common Assessment is made divided by the total annual Common Assessments for all Sites within the Master Association Area for such Assessment Year.
 - Special Assessments. In addition to Common Assessments, the Board may, 8.9 subject to the provisions of this Section, levy one or more additional Assessments for the purpose of raising funds, not provided by Common Assessments, to: (a) construct or reconstruct, repair, remodel, replace or maintain improvements upon Master Association Properties, including necessary personal property related thereto, and to pay ad valorem taxes and insurance on Master Association Properties; (b) add to the Master Association Properties; (c) provide for necessary facilities and equipment to offer the services authorized in this Declaration; or (d) repay any loan made to the Master Association to enable it to perform any duty or function authorized in this Declaration. Such Assessment shall be known as a "Special Assessment." The proportion of each Special Assessment to be paid by an Owner of a Site shall be equal to the percentage of the annual Common Assessment for such Site for The Assessment year in which the Special Assessment is made divided by the total annual Common Assessments for all Sites within the Master Association Area for such Assessment Year, The Master Association shall notify Owners in writing of the amount and purpose of any Special Assessment and of the manner in which, and the dates on which, any such Special Assessment is payable, and the Owners shall pay any such Special Assessment in the manner so specified.
 - 8.10 Reimbursement Assessments. Subject to the provisions hereof, the Board may levy an Assessment, known as a "Reimbursement Assessment," against any Owner(s) to reimburse the Master Association for any loss sustained by reason of the willful or negligent fallure of such Owner(s), or a Subassociation in which such Owner or Owners are members, to comply with this Declaration, any Supplemental Declaration, the Articles of Incorporation, the Bylaws or the Rules and Regulations which shall have resulted in the expenditure of funds by the Master Association to remedy a problem or to cause such compliance; provided, however, that the amount of any Reimbursement Assessments levied against such Owner's Subassociation shall be limited to a sum equal to the total of all such Reimbursement Assessments against such

Subassociation's members multiplied by a fraction, the numerator of which is the most recently assessed annual Common Assessments against such Owner and the denominator of which is the total of the most recently assessed annual Common Assessments against all Owners in such Subassociation. Any Relmbursement Assessment shall be levied only after Notice and Hearing. The amount of the Reimbursement Assessment shall be due and payable to the Master Association thirty days after notice to the Owner(s) of the decision of he Board stating that the Reimbursement Assessment is owning and stating the amount thereof.

- 8.11 Priority of Lien for Assessments. All sums assessed against any Site pursuant to this Declaration, together with court costs, reasonable attorneys' fees, late charges, and interest as provided herein, shall be secured by an equitable charge and continuing lien on such Site in favor of the Master Association. Such lien shall be superior to all other liens and encumbrances on such Site except only for: (a) liens of ad valorem taxes; and (b) a lien for all sums unpaid on a first Mortgage, on any secondary purchase money Mortgage, or on any Mortgage to Declarant, and all amounts advanced pursuant to any such Mortgage and secured thereby in accordance with the terms of such instrument; provided, however, that the subordination of the lien for Assessments to the foregoing Mortgages shall apply only to Assessments that shall have become due and payable prior to a sale or transfer of such Site pursuant to a Foreclosure. Such sale or transfer shall not relieve such Site from liability for any Assessments accruing after such sale or transfer. All Persons acquiring other liens or encumbrances on any Site after the Recording of this Declaration shall be deemed to consent that such liens or encumbrances shall be inferior to such future equitable charges and liens for Assessments as provided herein, whether or not such prior consent shall be specifically set forth in the Instruments creating such liens or encumbrances.
- Effect of Nonpayment of Assessment; Remedies of the Association. Assessments or any portion thereof not paid when due shall be delinquent. Any Assessment delinquent for a period of more than ten days shall incur a late charge in an amount as may be determined by the Board from time to time, shall bear interest from the date due at the rate of one and one-half percent (1 1/2%) per month until the date of payment, and the Board shall cause a notice of delinquency to be given to any Owner not paying within ten days following the due date. If any installment of an Assessment has not been paid within thirty days of the due date thereof, the entire unpaid balance of the Assessment may be accelerated at the option of the Board and, if so accelerated, shall thereupon become forthwith due and payable in full. The continuing lien and equitable charge of such Assessment shall include the late charge established by the Board, interest on the principal amount due at the rate of one and one-half percent (1 1/2%) per month until paid, all costs of collection (including reasonable attorneys' fees and court costs), and any other amounts provided or permitted hereunder or by law. If the Assessment remains unpaid after sixty days from the original due date, the Association may, as the Board shall determine, institute sult to collect such amounts to foreclose its lien or to do both. The equitable charge and lien provided for in this Section shall be in favor of the Master Association; and each Owner, by its acceptance of a deed to a Site, vests in the Master Association or its agents the right and power to sue or otherwise proceed against such Owner for the collection of such charges and/or to foreclose the Master Association's lien. The Master Association shall have the power to bid on the Site at any such foreclosure sale and to acquire, hold, lease, mortgage, and convey the Site except that the amount the Master Association may bid at any such sale may not exceed the total amount owed to the Master Association by the delinquent Owner.
- 8.13 <u>Exempt Property.</u> The following property and Persons subject to this Declaration shall be exempted from all Assessments hereunder: (a) the grantee in conveyances made for the purpose of granting utility easements; and (b) the Master Association and all Master Association Properties.
- 8.14 Estoppel Certificate. Upon the payment of such reasonable fee as may be determined from time to time by the Board and upon the written request of any Owner, any

Mortgagee or Person with, or intending to acquire, any right, title or interest in the Site of such Owner, the Master Association shall furnish a written statement setting forth the amount of any Assessments or other amounts, if any, due and owing to the Master Association and then unpaid with respect to such Site and/or the Owner thereof and setting forth the amount of any assessment levied against such Site which is not yet due and payable. Such statement shall, with respect to the Person to whom it is issued, be conclusive against the Master Association to establish that for all purposes that no greater or other amounts were then due or accrued and unpaid and that no other Assessments were then levied and unpaid against such Site.

8.15 No Offsets. All Assessments shall be payable in the amounts specified in the levy thereof, and no offsets or reductions thereof shall be permitted for any reason, including, without limitation, any claim of non-use of Master Association Properties or improvements or any claim of non-use of Master Association Properties or improvements or any claim that the Master Association, the Board of the Development Review committee is not properly exercising its duties and powers under this Declaration.

ARTICLE IX GENERAL RESTRICTIONS APPLICABLE TO PROPERTY

- 9.1 <u>Application of Restrictions</u>. All real property within the Master Association Area shall be held, used and enjoyed subject to the limitations, restrictions and other provisions set forth in this Declaration. Reasonable variances from the strict application of the limitations and restrictions in this Article IX in any specific case may be granted by the Board or the Development Review Committee in accordance with Section 10.25 hereof if such strict application would be unreasonable or unduly harsh under the circumstances; provided, however, that any such variance shall not materially injure any of the Master Association Areas. Any such variances must be in writing or be contained in written guidelines or rules promulgated by the Development Review Committee, and no such variance shall constitute a waiver or estoppel with respect to any future action by the Development Review committee or the Board.
- Maintenance of Property. No property within the Master Association Area shall be permitted to fall into disrepair, and all property within the Master Association Area, including any improvements and landscaping thereon, shall be kept and maintained by the Owners thereof in a clean, safe, attractive and sesthetically pleasing condition, not visibly in disrepair or lacking exterior paint, with windows glazed and paving swept. Any building on any Site which is vacant for any reason shall be kept locked and the windows shall be glazed in order to prevent entrance by vandals. All grounds shall be maintained in a safe, clean and neat condition free of rubbish and weeds. Lawns shall be kept in a mowed condition. Roads and pavements shall be kept reasonably true to line and grade in good repair. Drainage systems shall be kept clean and free of any obstacles. Parking areas, lighting and signage shall be kept clean and in good repair. To the extent reasonably possible, all plantings shall be maintained in a healthy condition, and fertilization, weeding and pruning of them shall be carried out on a regular basis. Dead or dying plants shall be removed as quickly as possible, and, subject, to seasonal limitations, replaced as quickly as possible, and in any event within thirty days after notification from the Development Review Committee. Impation systems shall be kept in proper working condition, with adjustment, repair and cleaning being done on a regular basis. Maintenance, repair and upkeep of each Site and any unpaved portion of the right-of-way of a public street adjoining such Site shall be the responsibility of the Owner of such Site. Maintenance, repair and upkeep of Master Association Properties shall be the responsibility of the Master Association. Violation of any provision of this Article IX or of this Declaration or of any applicable Supplemental Declaration by an Owner or by a Related User of an Owner shall be grounds for the Master Association, through the Board or agents, after Notice and Hearing, to enter upon the Site of the Owner and cure the violation or cause compliance with the provision or provisions and, as provided in Section 8.10 hereof, to levy and collect a Reimbursement Assessment for the costs and expenses of the Master Association in

so doing; provided, however that there shall be no entry into the interior of any building located on such Site without the consent of the Owner thereof unless an emergency exists.

- 9.3 Noxious or Offensive Activity. No noxious or offensive activity shall be carried on upon any property within the Master Association Area, nor shall anything be done or placed thereon which is or may become a nulsance or cause an unreasonable embarrassment, disturbance or annoyance to others.
- within the Master Association Area which is noxious or unreasonably offensive to others. Without limiting the generality of the foregoing, no exterior speakers, homs, whistles, bells or other sound devices, other than devices used exclusively for security, fire prevention or fire control purposes, shall be located or used on any property except with the prior written approval of the Development Review Committee.
- 9.5 <u>Hazardous Activities</u>. No activity shall be conducted on and no improvement shall be constructed on any property within the Master Association Area which is or might be unsafe or hazardous to any person or property. Without limiting the generality of the foregoing, no radiation or toxic substances shall be created on or emitted from any property, no firearms shall be discharged upon any property, and no open fires shall be lighted or permitted on any property except in a contained barbeque unit while attended and used for cooking purposes or within an interior or exterior fireplace designed to prevent the dispersal of burning embers.
- 9.6 Garbage and Trash. No refuse, garbage, trash or scrap lumber or metal; and no grass, shrub or tree clippings; and no plant waste, compost, bulk materials or debris of any kind shall be kept, stored or allowed to accumulate on any Site except within an enclosed structure or container approved by the Development Review Committee or unless appropriately screened from view, in a manner acceptable to the Development Review Committee, except that any refuse container containing such materials and approved by the Development Review Committee may be placed outside at such times as may be reasonably necessary to permit garbage or trash pickup.
- 9.7 <u>Temporary Structures</u>. No tent, shack, temporary structure or temporary building shall be placed upon any property within the Master Association Area except with the prior written consent of the Development Review Committee obtained in each instance, which consent cannot be unreasonably withheld or delayed.
- 9.8 Antennas. Aerials and Dishes. No exterior radio antenna or aerial, television antenna, aerial, dish or similar facility of any type shall be erected or maintained in the Master Association Area (specifically including any site) without the prior written approval of the Development Review Committee of the location of and screening for any such object.
- gisplay of any kind shall be erected or maintained anywhere within the Master Association Area so as to be evident to public view, except signs as may be approved in writing by the Development Review committee in accordance with the Design Guidelines promulgated by the Master Association for Cool Springs, which sets forth the uniform standards relating to dimensions, color, number, graphics, style and location of signs. The only types of signs permitted within the Master Association Area are signs giving identity to the project (known as "Cool Springs") established pursuant to this Declaration, or to a development on a Site or a building, building group, shopping center, firm or company, either by logo, name or both, signs which provide information or direction, traffic signs and temporary signs either to denote the firms involved in the construction or financing of a project or the presence of a project or building available for sale or lease, or real estate directional signs providing directions from major roadways to Sites which do not contain any commercial message or advertisement, but only the name, trademark or trade symbol, or address

of the development or Site to which directions are being provided. The addresses of all buildings in the Master Association Area shall be identified either on a free-standing identification sign or on the building itself by means of numbers large enough to be readily visible from the nearest public street. Each Owner shall be responsible for the maintenance of signs within such Owner's Site and shall replace lamps and repair damaged signs as promptly as reasonably possible after any such lamp ceases to operate or such damage occurs. No signs shall be located in any way which create a traffic or other hazard, obstruct any other sign, or restricts visibility for vehicular or pedestrian circulation or the views of surrounding buildings or the environment. No Identification signs shall be allowed within the right-of-way of a public street or in any other area not approved by the Development Review Committee. All signs, except traffic signs, shall be set back a minimum of ten feet from any property line of a Site. Signs shall not advertise a product or a service, except as may be approved in writing by the Development Review Committee. Signs may be illuminated by floodlighting; however, the specific type of lighting of any sign must be approved by the Development Review Committee. In addition, the following types of signs are specifically prohibited: Signs which flash, blink, or are animated in any way; any sign which, by reason of its shape, position or color, may be confused with a county-authorized traffic sign or signal; any sign that contains the wording "stop" or "danger" or any other similar wording that may confuse or mislead traffic; any internally lit awning, and any temporary sign attached to a tree or utility pole or attached to or painted on a building or any natural feature on a Site. All electrical conduits and wiring for signs must be concealed from view or placed below ground. No sign manufacturer's name, union label or other lettering shall be visible on any sign. No exposed lamps, globes, tubes, electrical raceways, electrical appurtenances or similar devices will be permitted on any sign. No sign may be mounted on any roof. No sign may have more than two sign faces.

- 9.10 Prohibited Uses. The conducting of the following operations, occupations, and uses shall not be permitted in the Master Association Area: (a) trailer courts; (b) labor camps; (c) junk yards; (d) commercial storage of building or construction materials; (e) distillation of bones; (f) junk yards; (d) commercial storage of building or construction materials; (e) distillation of bones; (f) dumping, disposal, incineration or reduction of garbage, sewage, dead animals or refuse; (g) fat rendering; (h) stockyard or slaughter of animals; (l) smelting of iron, tin, zinc or other ores; (l) refining of petroleum or of its products; (k) any new cemeteries or mausoleums; (l) jall, penal, detention or correction farms; (m) gasoline service stations except on Sites, if any, specifically designated by Declarant; (n) mining, quanying, drilling or exploring for or removing oil, gas, other hydrocarbons minerals, rocks, stones, gravel or earth; (o) temporary, portable or permanent sawmill; (p) community fair, (q) sewage treatment plant; (r) boarding or breeding kennels; (s) medical facilities permitting abortions; (l) construction contractor (other than office uses and operations, which are permitted); (u) sanitarium or convalescent home; (v) any uses otherwise prohibited under the current Zoning Code for the City of Franklin; (w) sale of fireworks or other explosives; (x) the raising or keeping of animals (other than household pets); (y) establishments selling "adult" type books or other materials; and (z) massage parlors.
 - 9.11 Maintenance of Drainage. The established drainage pattern over any property within the Master Association Area shall not be altered except as approved in writing by the Development Review Committee. Approval shall not be granted unless provision is made for adequate alternate drainage. The "established drainage pattern" shall mean the drainage pattern which exists at the time the overall grading of any property is completed and shall include any established drainage pattern shown on any plans approved by the Development Review Committee. The established drainage pattern may include the drainage pattern from Master Association Properties, or from any Site over another Site. This Section shall not apply to Declarant in connection with its development, construction and sale of property within the Master Association Area.
 - 9.12 Compliance with Insurance Requirements. Except as may be approved in writing by the Board, nothing shall be done or kept on property within the Master Association Area which

may result in a material increase in the rates of insurance maintained by the Master Association or which would result in cancellations of any insurance maintained by the Master Association.

- 9.13 Compliance with Laws. Nothing shall be done or kept on any property within the Master Association Area in violation of any law, ordinance, rule or regulation of any governmental authority having jurisdiction.
- 9.14 <u>Sewage Disposal Systems</u>. No casspool, septic tank or other sewage disposal system shall be installed within the Master Association Area without the prior written consent of the Development Review Committee, except central sewage disposal systems installed and maintained by a governmental sanitation agency providing commercial sewage disposal services. Any sewage disposal system installed for property within the Master Association Area shall be subject to applicable laws, rules and regulations of the governmental authority having jurisdiction.
- 9.15 <u>Water Systems</u>. No individual water supply system shall be installed or maintained for any property within the Master Association Area unless such system is approved in writing by the Development Review Committee and is designed, located, and constructed and equipped in accordance with the requirements, standards and recommendations of any applicable governmental authority having jurisdiction.
- 9.18 <u>Restoration in Event of Damage or Destruction</u>. In the event of the damage or destruction of any improvement on any Site, the Owner thereof shall, as soon as the extent of damage will permit (and, if reasonably possible, within thirty days after notification by the Development Review Committee), cause the damaged or destroyed improvement to be replaced or restored to its original condition or such other condition as may be approved in writing by the Development Review Committee, or the Owner shall cause the damaged or destroyed improvement to be demolished and the Site to be suitably landscaped, subject to the approval of the Development Review Committee, so as to present a pleasing and attractive appearance.
- Land Coverage. The aggregate floor area of all buildings to be constructed on any nonresidential Site within the Master Association Area shall be limited in accordance with any Supplemental Declaration applicable to such Site in order that the ratio of the floor area of all such buildings to the area of the Site on which they are constructed shall not exceed the maximum ratio of floor area to the Site area as set forth in any such Supplemental Declaration or the City of Franklin, Tennessee standards, whichever is the more restrictive. Moreover, the extent to which each Site shall be covered with improvements shall be limited so as to provide a minimum of thirty percent unobstructed space (landscape surface ratios) not covered with improvements on each nonresidential Site or on contiguous nonresidential Sites which are part of a coordinated development. Multifamily Residential Sites shall not be developed so as to contain (a) more than twelve dwelling units per acre (gross density), or (b) an aggregate floor area greater than thirty-five percent of the area of the site and a minimum of twenty-five percent of unobstructed space (landscape surface ratios) not covered improvements on each residential Site or on contiguous residential sites which are part of a coordinated development. Nevertheless, such unobstructed open space not covered by improvements may include any pedestrian pavements (sidewalks and plazas) passing through open space areas, drainageways, water bodies and detention and retention areas. Single Family Residential Sites zoned planned residential shall not be developed so as to contain an aggregate maximum building coverage, including accessory buildings, of more than forty percent of each site.
- 9.18 <u>Height of Rooftop Structures</u>. No structure or appurtenance, including but not limited to water towers, standplpes, penthouses, elevators, elevator equipment, stairways, ventilating fans or similar equipment required to operate and maintain any building, fire or parapet walls, skylights, tanks, cooling or other towers, wireless radio or television masts, satellite dishes,

or flagpoles shall exceed a height of ten feet above the finished rooftop of any building, except as may be specifically approved in writing by the Development Review Committee.

- 9.19 Parking Structures. Screen landscape, walls, or a combination of the two, at least three feet six inches tall, shall be provided at the perimeter of each parking level above grade. Any ramp driveway exit from a parking structure shall end not closer than twenty-five feet to any side property line, fifty feet from any front property line, and forty feet from any rear property line of a Site. Where the height of the structure exceeds thirty-five feet, all yards shall be increased by one foot for each foot of building height in excess of thirty-five feet. In addition, landscaping shall be provided on the surface level of parking structures in accordance with the requirements of the Development Review Committee and City of Franklin standards.
- Parking and Circulation. Vehicles shall be parked only in designated, paved parking spaces either on the Site served by such spaces or within 300 feet of such Site. Nonresidential parking areas shall be separated from any street right-of-way line by a landscaped strip thirty feet wide. In nonresidential areas, landscaped Islands shall be provided at maximum intervals of every twenty-four parking spaces. Said landscaped area shall be seven hundred twenty square feet in size and islands shall occur at the end of all rows of parking. In Residential areas, landscape islands shall be provided at maximum intervals of every ten parking spaces as well as at the end of all rows of parking. Parking Islands shall have a minimum width of nine feet, and a continuous curb shall be provided around parking Islands to prevent vehicular intrusion. Parking areas shall be divided into sections with capacities of not more than 200 cars each, and each section shall be separated by major landscaped buffers (minimum twenty feet wide) to provide visual relief. No parking areas or structures shall be permitted within buffer areas. The requirements for landscape islands may be waived in writing by the Development Review Committee, after review of the relevant landscape drawings, to the extent that an equivalent area of existing trees is preserved on the other areas of the site, and the Site meets all applicable local zoning ordinances.
- 9.21 <u>Curb Cuts</u>. Curb cuts for driveways vary according to the type of street on which the property fronts. The minimum Interval between permitted driveways on each lot for a local street is forty-five feet, for a collector seventy-five feet, for an arterial is two hundred ten feet and a frontage road is forty-five feet. Driveway curb cuts on each lot, measured from the centerline of each curb cut, shall be located no closer to a property line than twenty feet or one-third of the above-permitted interval, whichever is further, and no closer to an intersecting street right-of-way line than one-half of the above-permitted interval. Driveway curb cuts for single family residential lots shall be no less than five feet from any adjoining property. Other conditions shall be limited in accordance with any Supplemental Declaration.
- 9.22 <u>Pedestrian Circulation</u>. Buildings and parking areas shall be designed so that pedestrian access shall be provided from parking areas and public streets to building entries. All sidewalks shall have a minimum unobstructed width of five feet.
- 9.23 Loading and Servicing Areas. Loading doors, docks, material hauling facilities, accessory structures and servicing areas shall be adequately screened with landscaping or walls or a combination thereof so as not to be visible from any residential area or street and to minimize the effect of their appearance from neighboring nonresidential Sites. Moreover, loading and servicing areas shall be designed as an integral part of the building architecture and so that the entire loading and servicing operation can be conducted within the confines of any such area.
- 9.24 Storage Areas. No materials, supplies, merchandise, equipment, company-owned vehicles or similar items shall be stored except in an enclosed, covered building or on a Site in a location which shall have been adequately screened from the view of the adjacent Sites, public streets and pedestrian walkways by either a fence, wall or landscaping screen approved in writing

by the Development Review Committee. In addition, company-owned vehicles, trucks and equipment shall be stored only in areas so designated by a Supplemental Declaration or the Development Review Committee and shall be stored in an enclosed area meeting the requirements of the first sentence in this Section.

- 9.25 <u>Utilities</u>. Other than for street lighting, all pipes, lines and other facilities for utilities and all lines and conduits of any type hereafter installed for the transmission of audio and visual signals or electricity shall be located beneath the ground or within an enclosed structure, except that certain overhead lighting and utility appurtenances (including telephone pedestals, utility meters, imgation systems, back-flow preventors, transformers and switching devices) may be located above the ground if they are adequately screened by landscaping so as not to be visible from other Sites, streets, and pedestrian walkways in accordance with plans approved in writing by the Development Review Committee. To the extent possible, transformers, utility meters and other utility appurtenances shall be grouped together.
- Landscape Development. Every site on which a building is constructed and any unpaved portion of the right-of-way of any public street adjoining any such Site shall be landscaped by the Owner or Related User in accordance with plans and specifications prepared by a registered landscape architect and submitted to and approved in writing by the Development Review Committee. (Such approval is not in lieu of nor does it constitute an approval by the City of Franklin Site Plan Review process.) Furthermore, each Owner or Related User shall submit asbuilt drawings of all imigation systems to the Development Review Committee upon completion of the installation of any such system. Landscaping in accordance with the approved plan must be installed before the occupancy of any building within the phase of development to which such landscape plan pertains, except when seasonal limitations prohibit, in which case the landscaping must be installed within thirty days from the time planting operations can be feasibly undertaken. Moreover, when seasonal limitations do not permit planting, erosion control measures must be implemented in accordance with generally accepted practices in the real estate development industry, as approved by the Development Review Committee. The unpaved portion of the rightof-way of any public street and a thirty-foot wide strip of each Site adjoining each such unpaved portion shall be fully landscaped, and parking areas shall be landscaped so that, where possible, berming shall be provided to screen parked cars from the view of adjoining Sites and public streets. Where berms are not possible the parking areas shall be screened through the use of evergreen and coniferous trees, shrubs, screen walls, or any combination thereof, as approved by the Development Review Committee. A minimum of one tree shall be planted or allowed to remain for every 250 square feet of planting space in parking lot bay dividers, islands and medians in nonresidential areas. In addition, no fewer than twelve trees per acre must be provided for all surface parking lots and tops of parking structures in nonresidential areas. All areas of a Site which have been disturbed by land development activities and are not paved or covered with improvements or are not presently forested, must be landscaped with trees and imigated turf, except that shrub beds may be substituted judiciously and carefully. The Development Review Committee may waive this impation requirement for areas of future expansion of buildings and parking, after the DRC has reviewed relevant landscaping and architectural drawings, landscaping must be in accordance with City of Franklin standards. Large, uninterrupted areas of gravel, pine straw, bark mulch or bare soil are prohibited. In the absence of unusual extenuating circumstances approved in writing by the Development Review Committee, all landscaped areas (including the unpaved portion of the right-of-way of any public street) shall be impated by underground, fully automated inigation systems. Permanent inigation systems need not be provided for areas to be maintained as existing natural areas or areas to be restored as natural areas. Temporary imigation systems may be required by the Development Review Committee for the re-establishment of natural areas, though. Furthermore, undeveloped areas which are held in reserve for future buildings or pavement but which have been disturbed by land development activities need not be irrigated or fully landscaped as along as they are seeded with a turi mix to reduce weed growth and minimize erosion and as long as they are mowed and properly

maintained. Nevertheless, all unpaved public rights-of-way abutting a Site must be landscaped within two years of the purchase of a Site, whether or not all phases of the development on such Site are then complete, except that any such rights-of-way which shall have been irrigated and landscaped by Declarant or another prior Owner shall be maintained as irrigated, landscaped areas by the Owners responsible for the maintenance of such rights-of-way.

ARTICLE X DEVELOPMENT APPROVAL

- 10.1 Approval of Improvements Required. The approval of the Development Review Committee shall be required for any "Improvement to Property" (as defined in Section 10.2 hereof) on any Site, except for any Improvements to Property made by Declarant and except as prior approval may be waived or certain improvements to Property may be exempted in writing or under written guidelines or rules promulgated by the Board or the Development Review Committee because approval in such case or cases is not reasonably required to carry out the purposes of this Declaration.
- Improvement to Property. "Improvement to Property" requiring approval of the 10.2 Development Review Committee, shall include, but shall not be limited to: (a) the construction, installation, erection or expansion of any building, structure or other improvement, including parking areas and utility facilities; (b) the demolition or destruction, by voluntary action, of any building, structure or other improvement; (c) the grading, excavation, filling or similar substantial disturbance to the surface of the land, including, without limitation, change of grade, change of ground level, change of drainage pattern or change of stream bed; (d) the landscaping, planting, clearing or removing of trees, shrubs, grass or plants; (e) the placing, installing, modifying or erecting of any solar panels or other solar equipment, or of any antennas, aerials, microwave dishes, wires or boxes, or of any evaporative, cooling, air conditioning or heating equipment, or of any other similar items of equipment on the roof or exterior of any improvements or on any other portion of a Site which is visible from any part of another Site or from any part of the Master Association Area; (f) the erection, construction, removal, modification, substitution or remodeling of signs, fences, walls, patios, decks, planters or other similar improvements; and (g) any material, extenor change, remodeling or alteration of any previously approved improvement to Property, including any change of exterior appearance, color or texture.
- Review Committee shall consist of three members, and Declarant shall have the right to appoint such members during the Appointment Period. Upon the end of the Appointment Period, the Board shall meet and appoint the members of the Development Review Committee. The members of the Development Review Committee appointed by the Board may be removed and replaced at any time by the board and shall serve for such term as may be designated by the board or until their death or resignation or until their removal by the Board. After the Appointment Period, the Master Association may at any time, from time to time, change the authorized number of members of the Development Review Committee, but the number of members shall always be an odd number and shall not be less than three.
- 10.4 <u>Address of Committee</u>. Until changed by the Development Review Committee, the address of the Development Review Committee shall be at the principal office of Declarant in the State of Tennessee.
- 10.5 <u>Submission of Plans</u>. Subject to the provision of Section 10.1 hereof, prior to commencement of work to accomplish any proposed improvement to Property, the Person (the "Applicant") proposing to make such improvement to Property shall, in accordance with Sections

10.6 through 10.10 hereof and the guidelines promulgated by the Development Review Committee, submit to the Development Review Committee such descriptions, surveys, plot plans, drainage plans, elevation drawings, construction plans and specifications, building permits and samples of materials and colors as the Development Review Committee shall reasonably request showing the nature, kind, shape, height, width, color, materials, location and other essential features of the proposed improvements to Property. The Development Review Committee may require submission of additional plans, specifications or other information prior to approving or disapproving the proposed improvement to Property. Until receipt by the Development Review Committee of all required materials in connection with each phase of the review and approval procedure of the Development Review Committee for the proposed improvement to Property, the Development Review Committee may postpone review of any material submitted for approval.

- 10.6 <u>Pre-design Conference</u>. Prior to submitting preliminary plans for a proposed improvement to Property, the Applicant and, if possible, the Applicant's architect and other consultant(s) shall meet with the Development Review Committee for such committee to review and clarify to the Applicant applicable design guidelines of the Development Review Committee, their relation to the characteristics of the particular Site, and the technical issues related to review procedures.
- 10.6 hereof, the Development Review Committee shall meet to consider the preliminary plans to be presented to the Development Review Committee pursuant to such predesign conference. In accordance with the design guidelines of the Development Review Committee, the Applicant proposing the improvement to Property must submit to the Development Review Committee preliminary plans for the improvement to Property, which shall include the information for such improvement to Property required by the rules and guidelines of the Development Review Committee. Upon receipt of all required preliminary information, the Development Review Committee shall review it and respond to the Applicant in writing within the applicable time period specified in the design guidelines of the Development Review Committee.
- Development Review Committee to the Information submitted by Applicant pursuant to Section 10.7 hereof, the Applicant shall submit final design development plans for the Applicant's proposed Improvement to Property, which shall include the information required pursuant to the rules and guidelines of the Development Review Committee in addition to any revisions of the drawings submitted during the schematic design phase pursuant to Section 10.7 above. Following receipt of all such required plans, the Development Review Committee shall approve or disapprove in writing all such plans within the applicable time period specified in the design guidelines of the Development Review Committee.
- review is to assure the Development Review Committee that final construction plans comply with the design approved by the Development Review Committee. No construction of an improvement to Property may take place until the Development Review Committee shall have approved the final construction documents for such improvement. Upon the request of any Applicant whose design development plans for an Improvement to Property have been approved pursuant to Section 10.8 hereof, the Development Review Committee shall meet to consider such final construction documents. Thereafter, within the applicable time period specified in the design guidelines of the Development Review Committee, the Applicant must submit the final construction documents, plans, specification and other materials required by the rules and guidelines of the Development Review Committee. Following receipt thereof, the Development Review Committee shall approve or disapprove in writing all such documents, plans, specifications and materials within the applicable time period specified in the design guidelines of the Development Review Committee.

- Property, if an Owner or Related User desires to make a change in previously approved plans which affects the exterior of such improvement or the Site on which it is being built and which deviates from the plans as approved by the Development Review Committee, the owner or Related User must submit a written request to the Development Review Committee along with a "red-lined" set of plans clearly delineating the proposed change. The Development Review Committee shall review and respond to such request within the applicable time period specified in the design guidelines of the Development Review Committee. No changes from plans previously approved by the Development Review Committee may be made unless approved by the Development Review Committee.
- Criteria for Approval. The Development Review Committee shall approve any 10.11 proposed improvement to Property only if it deems, in its reasonable discretion, that: (a) the proposed improvement to Property in the location indicated will not be detrimental to the appearance of the surrounding areas of the Master Association Area as a whole; (b) the appearance of the proposed improvement to Property will be in harmony with the surrounding areas of the Master Association Area; (c) the proposed improvement to Property will not detract from the beauty, wholesomeness or attractiveness of the Master Association Area or the enjoyment thereof by Owners; (d) the proposed Improvement to Property is in compliance with the provisions of this Declaration, the applicable Rules and Regulations and the applicable guidelines and rules of the Development Review Committee; (e) the upkeep and maintenance of the proposed improvement to Property will not become a burden on the Master Association; (f) the plans for the proposed improvement to Property include such information as may have been reasonably requested by the Development Review Committee; (g) the exterior design, color scheme, finish, proportions, style of architecture, height, appearance and materials of the proposed improvement to Property are not objectionable; (h) the grading plan for the proposed improvement to Property is not objectionable; (i) the location and capacity of the parking areas within the proposed improvement to Property are satisfactory; and (1) no perceived conflict exists between the proposed improvement to Property and any applicable zoning, building, land use or other laws, ordinances, rules or regulations affecting the development, construction or use of the proposed improvement to Property. The Development Review Committee may condition its approval of any proposed improvement to Property upon the making of such changes therein as the Development Review Committee may deem appropriate. Approval by the Development Review Committee shall not be deemed to constitute compliance with the requirements of any local building codes, and it shall be the responsibility of the Applicant to comply therewith. If the work contemplated by any such plans and specifications shall require a building permit or other permit under local building codes, then the Applicant shall submit a copy of any and all such permits to the Development Review Committee within ten days after the issuance of such permits.
- Committee Guidelines or Rules. From time to time, the Development Review Committee may issue guidelines or rules relating to the standards, procedures, materials to be submitted, and additional factors which will be taken into consideration in connection with the approval of any proposed improvement to Property. By way of Illustration and not limitation, such guidelines or rules may amplify restrictions set forth in this Declaration and establish requirements for (a) site planning, (b) minimum numbers and dimensions of parking stalls, (c) the design of parking areas, (d) the design and location of loading and service areas, (e) site grading and drainage, (f) fences, (g) signage, (h) architectural design, (i) landscaping, (j) lighting, (k) minimum setback distances for buildings and other structures from public streets and adjoining Sites, (i) rooftop design and appearance, and (m) construction practices. Such guidelines or rules may specify circumstances under which the strict application of limitations or restrictions under this Declaration will be waived or deemed waived in whole or in part because strict application would be unreasonable or unduly harsh under the circumstances. Such guidelines or rules may waive the requirement for approval of any one or more improvements to Property from the requirement

for approval, if such approval is not reasonably required to carry out the purposes of this Declaration.

- 10.13 Development Review Fee. The Development Review Committee may, in its guidelines or rules, provide for the payment of a reasonable fee to accompany each request for approval of any proposed improvement to Property. The Development Review Committee may provide that the amount of such fee shall be uniform for similar types of any proposed improvement to Property or that the fee shall be determined in any other reasonable manner.
- Decisions of Committee. Decisions of the Development Review Committee shall be made by the affirmative vote of a majority of its members. Notice of any decision of the Development Review Committee shall be given to the Applicant pursuant to the provisions of Section 11.6 hereof. If the time period in which the Development Review Committee is to make its decision would expire on a Saturday, Sunday or legal holiday, then the Development Review Committee shall have until the next following business day which is not a Saturday, Sunday or legal holiday in which to make its decision and give notice thereof to the Applicant. The decision of the Development Review Committee shall be in writing. If the decision is not to approve a proposed improvement to Property, the reason therefor shall be stated, and, upon the request of the Applicant, the Development Review Committee shall suggest revisions that meet the Development Review Committee's requirement, and shall otherwise make reasonable efforts (at no cost to the Master Association) to aid the Applicant in preparing a proposal that would be acceptable to the Development Review Committee. Any subsequent submission by the Applicant shall be reviewed and acted upon by the Development Review Committee as provided herein for a new submission. The decision of the Development Review Committee shall be transmitted to the Applicant at the address furnished by the Applicant to the Development Review Committee.
- 10.15 Appeal to Board. If the Development Review Committee denies, imposes conditions on, or refuses approval of a proposed improvement to Property, the Applicant may appeal to the board by giving written notice of such appeal to the Master Association and the Development Review Committee within thirty days after such denial, imposition of conditions, or refusal. The Board or a tribunal appointed pursuant to the Bylaws for Notice and Hearing, and the Board or such tribunal shall decide whether or not the proposed improvement to Property or the conditions imposed by the Development Review Committee shall be approved, disapproved or modified.
- 10.16 Failure of Committee to Act on Plans. Subject to the following sentence of this Section, any request for approval of a submission of plans and other materials for a proposed improvement to Property shall be deemed approved unless notice of disapproval is given to the Applicant by the Development Review Committee within the applicable time period for response by the Development Review Committee as set forth in the guidelines promulgated by it, unless such time period is extended by mutual agreement.
- improvement to Property, the proposed improvement to Property shall be commenced and completed as promptly and diligently as reasonably possible and in complete conformity with the description of the proposed improvement to Property, any materials submitted to the Development Review Committee In connection with the proposed improvement to Property and any conditions imposed by the Development Review Committee. Fallure to complete the proposed improvement to Property within eighteen months after the date of approval or to complete the improvement to Property In accordance with the description and materials furnished to, and the conditions Imposed by, the Development Review Committee, shall constitute noncompliance with the requirements for approval of the proposed improvement to Property.

- Applicant shall give the Development Review Committee written notice of the completion of the improvement to Property and a certificate from an architect licensed to practice in Tennessee stating that the improvement to Property has been substantially completed in accordance with the stating that the improvement to Property has been substantially completed in accordance with the stating that the improvement to Property has been substantially completed in accordance with the description and materials furnished to, and the conditions imposed by, the Development Review Committee (collectively the "Notice of Completion"). Until the date of receipt of the Notice of Completion, the Development Review Committee shall not be deemed to have notice of the completion of such improvement to Property.
- 10.19 Inspection of Work. The Development Review Committee or its duly authorized representative shall have the right to inspect any improvement to Property at any reasonable time or times prior to or after completion, provided that the right of Inspection shall terminate thirty days after the Development Review Committee shall have received a Notice of Completion from the Applicant.
- Development Review Committee finds that any improvement to Property has been made without obtaining the approval of the Development Review Committee or was not done in substantial compliance with the description and materials furnished to, and any conditions imposed by, the Development Review Committee, or was not completed within eighteen months after the date of approval by the Development Review Committee, the Development Review Committee shall notify the Applicant in writing of the noncompliance (the "Notice of Noncompliance"); which Notice of Noncompliance shall be given, if given at all, not later than thirty days after the Development Review Committee receives any applicable Notice of Completion from the Applicant. The Notice of Noncompliance shall specify the particulars of the noncompliance and shall require the Applicant to take such action as may be necessary to remedy the noncompliance, including removal of any portion of the improvement to Property which is not in compliance.
 - Applicant's act or neglect, the Development Review Committee falls to notify the Applicant of any noncompliance within thirty days after receipt by the Development Review Committee of a Notice of Completion from the Applicant, the improvement to Property shall be deemed in compliance if the improvement to Property was, in fact, completed as of the date of Notice of Completion.
 - Committee gives any Notice of Noncompliance, the Applicant may appeal to Board by giving written notice of such appeal to the Board and the Development Review Committee within thirty days after the notice of Noncompliance was given by the Development Review Committee to the Applicant. If, after a Notice of Noncompliance, the Applicant fails to commence action diligently to remedy such noncompliance, the Development Review Committee shall request a finding of noncompliance by the Board by giving written notice of such request to the Master Association and the Applicant. In either event, the Board or a tribunal appointed pursuant to the Bylaws shall hear the matter in accordance with the provisions of the Bylaws for Notice and Hearing, and the Board shall decide whether or not there has been such noncompliance and, if so, the nature thereof and the estimated cost of correcting or removing the same.
 - 10.23 Correction of Noncompliance. If the Board determines that a circumstance of noncompliance exists, the Applicant shall remedy or remove the same within a period of not more than forty-five days from the date the Board gives the Applicant notice of such determination by the Board or applicable tribunal. If the Applicant does not comply with the ruling within such period, the board may, at its option, Record a final Notice of Noncompliance against the Site on which the noncomplying improvement to Property exists, shall be entitled to remove such noncomplying improvement to Property or otherwise to remedy the noncompliance such as by having construction enjoined or maintaining an action for the removal of such noncomplying

improvement to Property, and the Applicant shall relimburse the Master Association, upon demand, for all expenses incurred in connection therewith. If such expenses are not promptly repaid by the Applicant or Owner to the Master Association, the board may levy a Reimbursement Assessment against the Owner of the Site for such costs and expenses. The right of the Master Association to remedy or remove any noncomplying Improvement to Property shall be in addition to all other rights and remedies which the Master Association may have at law, in equity or under this Declaration.

- Review Committee or by the Board shall constitute a waiver or estoppel with respect to future action by the Development Review Committee or the Board with respect to any improvement to Property. Specifically, the approval of the Development Review Committee of any improvement to Property shall not be deemed a waiver of any right or an estoppel to withhold approval or consent for any similar or dissimilar improvement to Property or any similar or dissimilar proposals, plans, specifications or other materials submitted with respect to any other improvement to Property by the same or any other Applicant.
- Committee Power to Grant Variances. The Development Review Committee 10.25 may authorize reasonable variances from strict compliance with any of the provisions of Articles IX and X hereof or (if so authorized by the applicable Supplemental Declaration) any development restrictions of any Supplemental Declaration, including restrictions upon height, size, floor area or placement of structures or similar restrictions, when circumstances such as topography, natural obstructions, aesthetic or environmental considerations may require, or when strict application would be unreasonable or unduly harsh under the circumstances; provided, however, that no such variance shall materially adversely affect the value of any of the Master Association Area. Such variances must be in writing or be contained in written guidelines or rules promulgated by the Development Review Committee and shall become effective when signed by a majority of the members of the Development Review Committee. If any such variance is granted, no violation of the provisions of this Declaration or (if so provided in the applicable Supplemental Declaration) any Supplemental Declaration shall be deemed to have occurred with respect to the matter for which the variance was granted; provided, however, that the granting of a variance shall not operate to waive any of the provisions of this Declaration or any Supplemental Declaration for any purpose except as to the particular property, the particular improvement and the particular provision hereof covered by the variance, nor shall the granting of any variance affect the jurisdiction of either any development restrictions of a Subassociation or any development review committee authorized by a Supplemental Declaration, nor shall the granting of a variance affect in any way the Owner's obligation to comply with all governmental laws and regulations affecting the property concerned, including, but not limited to, zoning ordinances and setback lines or other requirements imposed by any governmental authority having jurisdiction.
 - 10.26 Compensation of Members. Members of the Development Review Committee shall be entitled to receive reasonable compensation from the Master Association for services performed as members of such committee and to be reimbursed for out-of-pocket expenses incurred by them in the performance of their duties hereunder, provided, however, that any such member shall receive such compensation and reimbursement only from Declarant (if so agreed by the Declarant) and/or the fees paid to the Development Review Committee by Applicants.
 - 10.27 Committee Representative. The Development Review Committee shall meet from time to time as necessary to perform its duties hereunder. The Development Review Committee may, from time to time, by a resolution in writing adopted by a majority of the members, designate any one or more Persons as the representative of the Development Review Committee (the "Development Review Committee Representative") to take any action or perform any duties for or on behalf of the Development Review Committee, except the granting of approval to any submission for an improvement to Property and the granting of variances. The

Development Review Committee Representative may, but need not, be a member of the Development Review Committee. The action of the Development Review Committee Representative within the authority of such Development Review Committee Representative or with the written consent or the vote of a majority of the members of the Development Review Committee shall constitute action of the Development Review Committee.

- 10.28 Records of Actions. The Development Review Committee shall report in writing to the Board all final action of the Development Review Committee, and the Board shall keep a permanent record of such reported action.
- determined from time to time by the board and upon the reasonable request of any interested party, the Board shall, after confirming any necessary facts with the Development Review Committee, furnish a certificate with respect to the approval or disapproval of any improvement to Property or with respect to whether any improvement to Property was made in compliance herewith. Any Person without actual notice to the contrary shall be entitled to rely on such certificate for all matters set forth therein.
- Development Review Committee, any member of the Development Review Committee, any Development Review Committee, any member of the Development Review Committee Representative, the Master Association, any member of the Board, any tribunal thereof or Declarant for any loss, damage or injury arising out of or in any way connected with the performance of the duties of the Development Review Committee, any Development Review Committee Representative, the Board or such tribunal hereunder unless due to the willful misconduct or bad faith of the party to be held liable. In reviewing any matter, the Board, the Development Review Committee, any Development Review Committee Board, the Development Review Committee, any Development Review Committee Representative and Declarant shall not be responsible for reviewing, nor shall its approval of an improvement to Property be deemed approval of, the Improvement to Property from the standpoint of safety, whether structural, fire, security or otherwise, or as to conformance with building codes or other governmental laws or regulations.
- 10.31 Construction Period Exception. During the course of actual construction of any permitted structure or improvement to Property, and provided construction is proceeding with due diligence, the Development Review Committee shall temporarily suspend the provisions contained in the Declaration as to the property upon which the construction is taking place to the extent necessary to permit such construction; provided that, during the course of any such construction, nothing is done which will result in a violation of any of the provisions of this Declaration upon the completion of construction and nothing is done which will constitute a nuisance or unreasonable interference with the use and enjoyment of other property in the Master Association Area.
- 10.32 Appeal to Court of Law. No provision of this Article X shall predude the right of an aggreed Applicant to appeal any decision of the Development Review Committee or the Board to a court of law for a final decision, provided, however such aggreed Applicant must first have exhausted all of his or her remedies and appeal procedures, if any, as are authorized under this Declaration.
- Declaration which may be to the contrary, the provisions of this Article X shall not apply to any improvement to Property proposed or made by Declarant or its successors and assigns (as such term is defined in Section 2.17 hereof) in connection with its development, construction, promotion, marketing, sale or leasing of properties within the Annexable Area.

ARTICLE XI

MISCELLANEOUS PROVISIONS

- Master Association Properties, shall be binding upon and inure to the benefit of Declarant, the Association, and all Owners and Mortgagees, and their respective heirs, executors, legal representatives, successors, and assigns, and shall be and remain in effect until the twentieth anniversary of the date of the Recording of this Declaration, whereupon this Declaration shall expire automatically. Notwithstanding the foregoing, any easements granted pursuant hereto or in any Supplemental Declaration are and shall be perpetual, except to the extent, if any, otherwise provided in the creation of any such easement, and except that any dedication to an acceptance by an appropriate governmental authority or any conveyance or grant to any appropriate public utility of the facilities that are the subject of any such easements shall terminate those easements if such dedication, conveyance, or grant so provides. It is the Intent of Declarant that, upon the twentieth anniversary of the date of the Recording of this Declaration, a new set of protective covenants shall be drafted, adopted, and recorded by the Owners and the Master Association and that the Master Association shall not dissolve or terminate upon such twentieth anniversary date.
- 11.2 Land Outside Master Association Area. The restrictions created by this Declaration benefit and burden only the Master Association Area and no other land whatsoever, whether or not within the Annexable Area. Notwithstanding any possible sharing of present or future facilities by other land, whether developed by Declarant or others, the general plan created by the restrictions of this Declaration extends only to the Master Association Area, and such restrictions are not intended to benefit any Persons other than those having an interest in the Master Association Area. No Persons owning land or having an interest in land outside of the Master Association Area shall have any right whatsoever to enforce this Declaration for the benefit of such land.
- 11.3 Amendment to Declaration by Declarant. Until the first Site subject to this Declaration has been conveyed by Declarant to a non-Declarant Owner by a Recorded deed, any of the covenants, conditions, restrictions, easements, equitable servitudes and other provisions contained in this Declaration may be unilaterally amended or terminated by Declarant, or new covenants, conditions, restrictions, easements, reservations, rights-of-way, equitable servitudes and other provisions may be unilaterally added by Declarant, by the Recordation of a written instrument, executed by Declarant, setting forth such amendment, termination or additions.
- 11.4 Amendment to Declaration by Owners. Except as provided in Section 11.3 hereof, amendments to this Declaration shall be proposed and adopted in the following manner. Notice of the subject matter of the proposed amendment shall be included in the notice of the meeting of the Delegates at which such proposed amendment is to be considered and shall be delivered in accordance with the Bylaws. At such meeting, a resolution for the adoption of a proposed amendment may be proposed either by the Board or by a Delegate. Such amendment must be approved by at least sixty percent of the total votes entitled to be cast thereon; provided, however, that any amendment that (a) limits or curtails any easement rights granted or reserved herein, (b) delays the levying of any Assessment, (c) changes the basis of calculation of any Assessment, or (d) modifies the priority of the Master Association's lien for Assessments set forth in Section 8.11 hereof must be approved by the unanimous vote of all Delegates, and, as to an amendment to Section 8.11, all First Mortgagees, secondary purchase money Mortgagees, and the holder of any Mortgage executed in favor of Declarant. Furthermore, as long as Declarant owns any land within the Master Association Area, and during the Appointment Period no amendment to the Declaration may become effective without the approval of Declarant. The agreement of the required percentage of the Owners and, when required, the Declarant and any Mortgagee, to any amendment to this Declaration shall be evidenced by their execution of such amendment, or in the

alternative (as to all such Persons other than Declarant), the awom statement of the President and any Vice president or the Secretary of the association attached to or incorporated in the amendment executed by the Association, which swom statement shall state unequivocally that the written agreement of the required Persons was lawfully obtained. Any such amendment to this Declaration shall become effective only when Recorded, or at such later date as may be specified in the amendment itself.

- 11.5 <u>Association Right To Mortgage Information</u>. Each Owner hereby authorizes any Mortgagee holding a Mortgage on such Owner's Site to furnish information to the Master Association concerning the status of such Mortgage and the loan which it secures, to the extent such information is appropriate in order to assist the Master Association in determining if such loan is a valid First Mortgage or secondary purchase money Mortgage.
- 11.6 Notices. Upon any Owner's acquisition of a Site, such Owner shall notify the Master Association of such Owner's address for purposes of the furnishing of notice in connection with this Declaration. Until notice to the contrary is given to all Owners by the Master Association, the address of the Master Association for the purposes of the furnishing of notice in connection with this Declaration shall be Declarant's principal office in the State of Tennessee. Any notice permitted or required to be given under this Declaration shall be in writing and may be given either personally or by mail, telecopler, or telegraph. If served by mail, such notice shall be sent firstclass, postage prepaid, addressed to the Person entitled to receive such notice at the address given by such Persons to the Master Association (or, for the purposes of Article X of this Declaration, to the address given by such Persons to the Development Review Committee) for the purpose of service of such notice, or to the Site of such Person if no address has been given to the Master Association, and shall be deemed given, if not actually received earlier, at 5:00 PM on the second day which is not a Sunday or legal holiday after it is deposited in a regular depository of the United States Postal Service. Such address may be changed by any such Person from time to time by notice in writing to the Master Association. Notice to one or more Owners of a Lot shall constitute notice to all such Owners. It shall be the obligation of every Owner to notify the Secretary of the Master Association immediately in writing of any change in address. Any Person who becomes the Owner of a Site after the date on which notice is delivered personally or malled shall be deemed to have received such notice if received by such Owner's predecessor in title to such Site.
 - 11.7 Enforcement The Master Association, acting by the authority of the Board, and any Owner shall each have the right, acting alone or together, to enforce any or all of the covenants, conditions, restrictions, easements, reservations, rights-of-way, equitable servitudes and other provisions contained in this Declaration against any Site and the Owner thereof. The right of enforcement shall include the right to proceed, jointly or severally, at law or in equity to prevent any breach hereof and/or to recover damages for any breach of the terms of this Declaration, the Bylaws, or the Rules and Regulations. Such proceedings may include, without limitation, suits to restrain or enjoin such breaches, actions for damages resulting therefrom, and actions in equity against any particular Site to enforce any lien created by this Declaration. Damages shall not be deemed adequate compensation for any breach or violation of any provisions of this Declaration, the Bylaws, or the Rules and Regulations, and, accordingly, the Declarant, the Master Association, and each Owner shall be entitled to relief by way of injunction, as well as any other remedy either at law or in equity. In addition, upon the violation of this Declaration, the Bylaws or the Rules and Regulations, and after Notice of Hearing, the Board shall have the power and authority (a) to Impose reasonable monetary fines which shell constitute a lien upon the Site of the Owner or Owners who are guilty of such violation (b) to suspend the right of such Owner(s) and Related User(s) who are guilty of such violation to vote in the Master Association; and the Board shall have the power to impose either or both of these sanctions. Any such suspension of rights may be for the duration of the infraction and for any additional period thereafter, not to exceed thirty days.

- 11.8 No Obligation to Enforce Declaration. Neither Declarant nor the Master Association, or its directors, officers or committees, shall be under any obligation to take any action to enforce the terms of this Declaration, the Bylaws, or the Rules and Regulations.
- 11.9 <u>Violations Constitute a Nuisance</u>. Any violation of any covenant, condition, restriction, easement, reservation, right-of-way, equitable servitude or other provision contained in this Declaration, whether by act or omission, is hereby declared to be a nuisance and may be enjoined or abated, whether or not the relief sought is for negative or affirmative action, by any Person entitled to enforce the provisions of this Declaration.
- 11.10 Enforcement by Self-Help. Declarant or the Master Association, or any authorized agent of either of them, may use self-help to enforce any of the covenants, conditions, restrictions, easement, reservations, rights-of-way, equitable servitudes, or other provisions contained in this Declaration, provided such self-help is preceded by Notice and Hearing as set forth in the Bylaws, unless an emergency exists.
- 11.11 <u>Violation of Law.</u> Any violation of any federal, state, municipal or local law, ordinance, rule or regulation pertaining to the ownership, occupation or use of any property within the Master Association Area is hereby declared to be a violation of this Declaration and shall be subject to any and all of the enforcement procedures set forth in this Declaration.
- 11.12 No Waiver. No delay or failure on the part of the Association or any other aggreed party to invoke any available right, power, or remedy in respect to a breach of this Declaration, the Bylaws, or the Rules and Regulations shall be held to be a waiver by that party of (or stop that party from asserting) any right, power, or remedy available to it upon the recurrence or continuance of such breach or the occurrence of a different breach.
- 11.13 Remedies Cumulative. The rights, powers, and remedies provided in this Declaration shall be cumulative and not restrictive of other remedies at law or in equity, and the exercise by a Person of any particular right, power, or remedy shall not be deemed an election of remedies or to preclude such Person's resort to other rights, powers, or remedies available to it.
- 11.14 Costs and Attomey's Fees. In any action or proceeding under this Declaration, the prevailing party shall be entitled to recover its costs and expenses in connection therewith, including reasonable attomeys' fees.
- 11.15 Limitation on Liability. The Master Association, the Board, the Development Review Committee, the Development Review Committee Representative, Declarant, any Delegate and any member, agent or employee of any of them shall not be liable to any Person for any mistake of judgment, whether negligent or otherwise, or for any action or any failure to act under this Declaration or any Supplemental Declaration If the mistake, action or failure to act was in good faith and without malice. In addition, the Board and the officers of the Master Association shall have no personal liability with respect to any contract or other commitment made by them, in good faith, on behalf of the Master Association (except to the extent that such directors or officers may also be Owners) and the Association, as an Administrative Function, shall indemnify the Board and such officers against, and hold, save and defend them free and harmless from any and all expense, loss, or liability to others on account of any such contract or commitment. In addition, each member of the Board or the Development Review Committee, each Delegate, any Development Committee Representative, and each officer of the Master Association shall be indemnified and held harmless by the Association, as an Administrative Function, from any expense, loss or liability to others by reason of having served in such capacity, against all expenses, losses, and liabilities, including court costs and reasonable attorneys' fees, incurred by or imposed upon such Person in connection with any proceeding to which he may be a party or may have become involved by reason of holding such position, whether or not he holds such

position at the time such expenses are incurred, except in cases in which the expenses, losses, and liabilities arise from a proceeding in which such Person is adjudicated guilty of willful misfeasance or malfeasance, misconduct, or bad faith in the performance of his duties. In the event of a settlement of any such proceedings, the indemnification provided hereby shall apply only when the Board approves such settlement and reimbursement as being in the best interests of the Master Association. Any right of indemnification provided in this Section shall not be exclusive of any other rights to which an indemnitee may be entitled.

- 11.18 General Development Information. Any brochures, maps, models, handouts, schematics, plans and facilities provided or available in connection with Declarant's development, construction, promotion, marketing, sale or leasing of property or improvements are provided for general information purposes only, are subject to change and deletion without notice by Declarant, by public or governmental authorities and by others and shall not obligate Declarant to develop, construct, promote, market, sell or lease any such property or improvements whatsoever or in any particular manner.
- 11.17 Use of "Cool Springs" Name. In order to avoid confusion in the minds of the public about the identify of any Person conducting a business in the Master Association Area, no Owner or Related User may use the name "Cool Springs" in the name of any building, development or other business conducted within the Master Association Area without the prior written consent of Declarant, which consent shall be granted, withheld or conditioned as Declarant may choose in its sole discretion.
- 11.18 Governing Law. This Declaration shall be construed and governed under the enforced in accordance with the laws of the State of Tennessee.
- 11.19 Captions for Convenience. The table of contents, titles, headings and captions used in this Declaration are intended solely for convenience of reference and shall not be considered in construing any of the provisions contained herein.
- 11.20 Interpretation. The Board shall have the right, power and authority to determine all questions arising under or in connection with this Declaration and to construe and interpret its provisions, and any determination, construction, or interpretation made in good faith by the Board shall be binding on all Owners. In all cases, the provisions set forth or provided for in this Declaration shall be construed together and given that interpretation or construction which, in the opinion of the Board, will best effect the understanding of the intent of the general purposes of this Declaration. The provisions hereof shall be liberally interpreted to effectuate the purposes set forth herein, and, if necessary, they shall be so extended or enlarged by implication as to make them fully effective.
- 11.21 Conflicts in Legal Documents. In case of conflicts between the provisions in this Declaration and the Articles of Incorporation, the Bylaws or the Rules and Regulations, this Declaration shall control. In case of conflict in the provisions in the Articles of Incorporation and the Bylaws, the Articles of Incorporation shall control.
- 11.22 Number Gender and Grammar. Unless the context requires a contrary construction, the singular wherever used herein shall be construed to mean the plural when applicable, and the necessary grammatical changes required to make the provisions hereof apply either to corporations or other entities or to individuals, men or women, shall in all cases be assumed as though in each case fully expressed.
- 11.23 <u>Severability</u>. Whenever possible, each provision of this Declaration shall be interpreted in such manner as to be effective and valid, but if the application of any provision of this Declaration to any Person or to any property shall be prohibited or held invalid, such

prohibition or invalidity shall not affect any other provision or the application of any provision which can be given effect without the valid provision or application, and to this end the provisions of this Declaration are declared to be severable.

- 11.24 Notice of Sale, Lease, or Mortgage. If an Owner sells, leases, grants a Mortgage on, or otherwise disposes of all or substantially all of a Site or the improvements thereon, such Owner must promptly furnish to the Master Association in writing the name and address of such Owner's purchaser, lessee, Mortgagee, or transferee. Nevertheless, no such notice shall be required for leases of space or individual units in a multi-tenant facility on a Site.
- 11.25 No Reverter. No covenant or restriction herein is intended to be or shall be construed as a condition subsequent intended to be or shall be construed as a condition subsequent or conditional limitation, or as creating a possibility of reverter.
- 11.26 Effective Date Of Declaration. The effective date of this Declaration shall be the date of its Recording.

Cool Springs Real Estate Associates, LP., a Tennessee Limited Partnership

By: Provident Life and Accident Insurace Company, a Tennessee Corporation,
Managing General Partner

Charles L Frazier, Special Assets Director

STATE OF TENNESSEE)
COUNTY OF DAwning

Before me, Kennett L. Campbell , a Notary Public for the State and County aforesaid, personally appeared Charles L. Frazier, with whom I am personally acquainted (or proved to me on the basis of satisfactory evidence), and who, upon oath, acknowledged himself to be the Special Assets Director of Provident Life and Accident Insurance Company, a Tennessee corporation that is the managing general partner of Cool Springs Real Estate Associates, L.P., the within named bargainor, a Tennessee limited partnership, and that he, as Special Assets Director of the managing general partner, being authorized so to do, executed the foregoing instrument for the purposes contained by signing the name of the limited partnership by himself as such Special Assets Director of the managing general partner.

WITNESS my hand and seal, this 4 the day of October, 1984.

Notary Public

My Commission Expires: May 25, 1996

Tract 206

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 206):

Beginning at iron pin set on the easterly right-of-way of Liberty Road, a frontage road on the east side of I-65, and the southerly right-of-way of Gillespie Lane; thence with the southerly right-of-way of Gillespie Lane the following calls:

- 1. South 81° 56' 38" East, 372.87 feet to an iron pin set; thence,
- 2. South 86° 01' 53" East, 395.33 feet to an iron pin set; thence,
- 3. South 82° 31' 09" East, 665.48 feet to an iron pin set on the westerly right-of-way of Carothers Road; thence with the westerly right-of-way of Carothers Road,
- 4. Southerly, 551.34 feet along the arc of a curve to the right having a radius of 5564.14 feet, a central angle of 05° 40′ 38″, and a chord bearing and distance of South 20° 09′ 30″ West, 551.11 feet to an iron pin set; thence,
- 5. South 22° 57' 48" West, 70.62 feet to an iron pin set; thence leaving said right-of-way,
- 6. North 88° 11' 08" West, 227.79 feet to an existing iron pin; thence,
- 7. North 79° 56' 40" West, 112.76 feet to an existing iron pin; thence,
- 8. South 05° 49' 04" East, 88.82 feet to an existing iron pin; thence,
- 9. South 75° 37' 33" West, 182.87 feet to an existing iron pin; thence,
- 10. South 80° 48' 04" West, 172.33 feet to an iron pin set; thence,
- 11. South 24° 51' 07" West, 21.57 feet to an existing iron pin; thence,
- 12. South 78° 37' 09" West, 355.66 feet to an existing iron pin; thence,
- 13. South 56° 00' 34" West, 321.97 feet to an existing iron pin; thence,
- 14. South 70° 07' 05" West, 128.21 feet to an existing iron pin; thence,
- 15. South 80° 03' 05" West, 200.33 feet to an existing iron pin on the easterly right-of-way of Liberty Road; thence with said right-of-way,
- 16. North 16' 04' 51" East, 1250.28 feet to an iron pin set; thence,
- 17. North 66° 13' 37" East, 73.54 feet to the POINT OF BEGINNING and containing 29.9846 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Resolution Trust Corporation as Receiver for Metropolitan Federal Savings and Loan Association, F.A., of record in Book 1121, Page 315, Register's Office for Williamson County, Tennessee.

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 207):

Beginning at iron pin set at the southeasterly radius return at the intersection of the of the southerly right-of-way of Cool Springs Boulevard and the westerly right-of-way of Carothers Road; thence with the westerly right-of-way of Carothers Road the following calls:

- 1. South 16' 15' 44" West, 682.95 feet to an iron pin set; thence,
- 2. Southerly, 82.10 feet along the arc of a curve to the right having a radius of 5564.14 feet, a central angle of 00° 50′ 43″, and a chord bearing an distance of South 16° 41′ 06″ West, 82.10 feet to an iron pin set in the centerline of Gillespie Lane; thence with the centerline of Gillespie Lane the following calls:
- 3. North 82° 41' 48" West, 659.40 feet to an iron pin set; thence,
- 4. North 86° 10' 48" West, 395.89 feet to an iron pin set; thence,
- North 82° 26' 48" West, 376.95 feet to an iron pin set on the easterly right-of-way of I-65; thence with the easterly right-of-way of I-65 the following calls:
- 6. North 07° 33' 12" East, 24.99 feet to an iron pin set; thence,
- 7. North 26° 33' 30" West, 75.98 feet to an existing concrete monument; thence,
- 8. North 16' 16' 16" East, 485.03 feet to an existing concrete monument; thence,
- 9. North 24° 06' 41" East, 105.58 feet to an existing iron pin; thence,
- 10. Northeasterly, 504.02 feet along the arc of a curve to the right having a radius of 470.87 feet, a central angle of 61° 19' 45", and a chord bearing and distance of North 62° 18' 57" East, 480.30 feet to an existing concrete monument on the southerly right-of-way of Cool Springs Boulevard; thence with the southerly right-of-way of Cool Springs Boulevard the following calls:
- 11. South 79° 38' 36" East, 190.16 feet to an iron pin set; thence,
- 12. South 76° 05' 04" East, 62.59 feet to an existing concrete monument; thence,
- 13. South 78° 06' 53" East, 198.47 feet to an existing concrete monument; thence,
- 14. South 73° 42' 19" East, 349.93 feet to an existing concrete monument; thence,
- 15. South 81° 26' 34" East, 151.51 feet to an existing concrete monument; thence,
- 16. South 73° 22' 49" East, 106.93 feet to an iron pin set at the northwesterly radius return at the intersection of the southerly right-of-way of Cool Springs Boulevard and the westerly right-of-way of Carothers Road; thence,
- 17. Southeasterly, 78.23 feet along the arc of a curve to the right having a radius of 50.00 feet, a central angle of 89° 38' 43", and a chord bearing and distance of South 28° 33' 37" East, 70.49 feet to the POINT OF BEGINNING and containing 29.5895 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Gary T. Baker, Trustee, of record in Book 1121, Page 345, Register's Office for Williamson County, Tennessee.

Tract 208

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 208):

Beginning at iron pin set at the intersection of the centerline of Gillespie Lane and the centerline of Liberty Road; thence with the centerline of Gillespie Lane,

1. North 82° 21' 01" West, 295.97 feet to an iron pin set on the easterly right-of-way of Carothers Road; thence with said easterly right-of-way,

2. Northerly, 65.34 feet along the arc of a curve to the left having a radius of 5670.14 feet, a central angle of 00° 39′ 37″, and a chord bearing and distance of North 16° 35′ 33″ East, 65.34 feet to an iron pin set; thence,

3. North 16° 15' 44" East, 681.66 feet to an iron pin set; thence,

- 4. Northeasterly, 78.85 feet along the arc of a curve to the right having a radius of 50.00 feet, a central angle of 90° 21' 27", and a chord bearing an distance of North 61° 26' 27" East, 70.93 feet to an iron pin set on the southerly right-of-way of Cool Springs Boulevard; thence,
- North 16° 37' 11" East, 129.92 feet to an iron pin set on the northerly right-of-way of Cool Springs Boulevard; thence,
- 6. South 73° 22' 49" East, 135.47 feet to an iron pin set in the centerline of Liberty Road; thence with said centerline the following calls:
- 7. South 09° 34′ 54" West, 110.50 feet to an iron pin set; thence,

8. South 09° 00' 10" West, 191.14 feet to an iron pin set; thence,

9. South 09° 32' 21" West, 586.44 feet to the POINT OF BEGINNING and containing 4.8428 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Gary T. Baker, Trustee, of record in Book 1121, Page 345, Register's Office for Williamson County, Tennessee.

Tract 209

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 209):

Beginning at iron pin set at the intersection of the southerly right-of-way of Cool Springs Boulevard and the centerline of Liberty Road; thence with the southerly right-of-way of Cool Springs Boulevard the following calls:

- 1. North 73° 22' 49" West, 449.25 feet to a concrete highway monument; thence,
- 2. North 65° 59' 12" West, 196.47 feet to a concrete highway monument; thence,
- 3. North 67° 24' 36" West, 261.42 feet to a concrete highway monument; thence,
- 4. North 27° 48' 50" West, 101.69 feet to a concrete highway monument on the easterly right-of-way of I-65; thence with said right-of-way the following calls:
- 5. North 05° 46' 39" West, 187.89 feet to an iron pin set; thence,
- 6. Northwesterly, 221.91 feet along the arc of a curve to the left having a radius of 622.95 feet, a central angle of 20° 24' 35", and a chord bearing and distance of North 11° 42' 40" West, 220.73 feet to an iron pin set; thence,
- 7. North 30° 23' 00" West, 275.96 feet to an iron pin set; thence,
- 8. North 33° 23' 52" West, 42.10 feet to an iron pin set; thence,
- 9. North 31° 40' 25" West, 194.23 feet to an iron pin set; thence,
- 10. Northwesterly, 406.96 feet along the arc of a curve to the right having a radius of 1029.35 feet, a central angle of 22° 39' 08", and a chord bearing and distance of North 16° 49' 10" West, 404.32 feet to an iron pin set; thence,
- 11. South 81° 52' 57" East, 966.34 feet to an existing iron pin; thence,
- 12. South 06° 36' 05" West, 552.47 feet to an existing iron pin; thence,
- 13. South 81° 57' 35" East, 584.38 feet to an iron pin set in the centerline of Liberty Road; thence with said centerline the following calls:
- 14. South 07° 30' 05" West, 411.70 feet to an iron pin set; thence,
- 15. South 08° 34' 51" West, 141.30 feet to an iron pin set; thence,
- 16. South 09° 34' 54" West, 316.93 feet to the POINT OF BEGINNING and containing 29.1978 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Gary T. Baker, Trustee, of record in Book 1121, Page 345, Register's Office for Williamson County, Tennessee.

Tract 210

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 210):

Beginning at iron pin set at the intersection of the southerly right-of-way of Gillespie Lane and the easterly right-of-way of Carothers Road; thence with the southerly right-of-way of Gillespie Lane,

- 1. South 82° 31' 09" East, 306.39 feet to an existing iron pipe; thence leaving said right-of-way,
- 2. South 10° 01' 55" West, 198.84 feet to an iron pin set; thence,
- 3. South 78° 15' 23" East, 198.37 feet to an iron pin set; thence,

- 4. North 16° 34' 51" East, 216.37 feet to an iron pin set on the southerly right-of-way of Gillespie Lane; thence with said southerly right-of-way,
- 5. South 82° 20' 16" East, 265.85 feet to an iron pin set; thence leaving said southerly right-of-way,
- 6. South 07° 21' 17" West, 600.98 feet to an existing iron pipe; thence,
- 7. North 72° 16' 15" West, 101.77 feet to an iron pin set; thence,
- 8. North 60° 16' 38" West, 294.76 feet to an existing iron pin; thence,
- 9. South 89° 26' 30" West, 44.22 feet to an existing iron pin; thence,
- 10. South 80° 45' 55" West, 48.01 feet to an existing iron pin; thence,
- 11. South 64° 25' 43" West, 121.55 feet to an existing P.K. Nail; thence,
- 12. North 88° 11' 08" West, 368.19 feet to an iron pin set on the easterly right-of-way of Carothers Road; thence with said easterly right-of-way,
- 13. North 22° 57' 48" East, 29.55 feet to an iron pin set; thence,
- Northeasterly, 580.22 feet along the arc of a curve to the left having a radius of 5670.14 feet, a central angle of 05° 51' 47", and a chord bearing and distance of North 20° 03' 56" East, 579.97 feet to the POINT OF BEGINNING and containing 9.7308 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Resolution Trust Corporation as Receiver for Metropolitan Federal Saving and Loan Association, F.A., of record in Book 1121, Page 315, Register's Office for Williamson County, Tennessee.

Tract 211

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993; Job No. 85-132, Work Order 4290 (Tract 211):

Commencing at iron pin set at the intersection of the southerly right-of-way of Gillespie Lane and the easterly right-of-way of Carothers Road; thence with the southerly right-of-way of Gillespie Lane, South 82°31'09" East, 306.39 feet to an existing iron pipe at the POINT OF BEGINNING of the herein described parcel; thence,

- 1. South 82° 35' 15" East, 223.19 feet to an iron pin set; thence leaving said right-of-way,
- 2. South 16' 34' 51" West, 216.37 feet to an iron pin set; thence,
- 3. North 78° 15' 23" West, 198.37 feet to an iron pin set; thence,
- 4. North 10° 01' 55" East, 198.84 feet to the POINT OF BEGINNING and containing 0.9998 acres or 43550 square feet, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Cool Springs Venture, of record in Book 1121, Page 362, Register's Office for Williamson County, Tennessee.

Tract 212

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 212):

Beginning at iron pin set at the intersection of the centerline of Gillespie Lane and the centerline of Liberty Road; thence with the centerline of Liberty Road,

- 1. North 09° 32' 21" East, 586.44 feet to an iron pin set; thence leaving said centerline,
- 2. South 82° 28' 05" East, 1162.03 feet to an existing iron pin; thence,
- 3. South 19° 00' 07" West, 607.00 feet to an iron pin set in the centerline of Gillespie Lane; thence with said centerline the following calls:
- 4. North 81° 35' 01" West, 565.99 feet to an iron pin set; thence,
- 5. North 82° 20' 16" West, 266.02 feet to an iron pin set; thence,
- 6. North 82° 34' 31" West, 223.28 feet to an iron pin set; thence,
- 7. North 83° 34' 08" West, 6.65 feet to the POINT OF BEGINNING and containing 15.0273 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Resolution Trust Corporation as Receiver for Metropolitan Federal Savings and Loan Association, F.A., of record in Book 1121, Page 308, Register's Office for Williamson County, Tennessee.

Tract 213

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 213):

Commencing at iron pin set at the intersection of the centerline of Gillespie Lane and the centerline of Liberty Road; thence with the centerline of Gillespie Lane the following calls: South 83° 34' 08" East, 6.65 feet; South 82° 36' 14" East, 229.93 feet; thence, South 82°20'16" East, 266.02 feet to an iron pin set at the POINT OF BEGINNING of the herein described parcel; thence,

- 1. South 81° 35' 01" East, 565.99 feet to an iron pin set; thence leaving said centerline,
- 2. North 19° 00' 07" East, 607.00 feet to an existing iron pin; thence,
- 3. North 18° 25' 21" East, 193.34 feet to an iron pin set; thence,

- 4. North 82° 24' 36" West, 1193.66 feet to an iron pin set in the centerline of Liberty Road; thence with said centerline the following calls:
- 5. North 09° 34' 54" East, 427.43 feet to an iron pin set; thence,
- 6. North 08° 34' 51" East, 141.30 feet to an iron pin set; thence,
- 7. North 07° 30' 05" East, 670.00 feet to an iron pin set; thence leaving said centerline,
- 8. South 84° 34' 13" East, 271.96 feet to an iron pin set; thence,
- 9. South 82° 51' 44" East, 381.38 feet to an iron pin set; thence,
- 10. South 82° 13' 49" East, 717.69 feet to an iron pin set; thence,
- 11. South 75° 37' 52" West, 124.94 feet to an iron pin set; thence,
- 12. South 44° 40' 15" West, 177.54 feet to an iron pin set; thence,
- 13. South 82° 13' 24" West, 274.02 feet to an iron pin set; thence,
- 14. South 19° 23' 30" East, 51.66 feet to an iron pin set; thence,
- 15. North 81° 19' 13" East, 263.86 feet to an iron pin set; thence,
- 16. North 73° 03' 32" East, 264.56 feet to an existing iron pin; thence,
- 17. South 18° 37' 45" West, 249.53 feet to an existing iron pin; thence,
- 18. South 19° 21' 53" West, 282.39 feet to an iron pin set; thence,
- 19. South 77° 03' 03" East, 956.75 feet to an iron pin set; thence,
- 20. South 75° 43' 57" East, 207.60 feet to an iron pin set; thence,
- 21. South 78° 29' 58" East, 258.41 feet to an iron pin set; thence,
- 22. South 76° 28' 03" East, 201.01 feet to an iron pin set; thence,
- 23. South 74° 41' 28" East, 180.46 feet to an iron pin set; thence,
- 24. South 80° 36' 58" East, 295.36 feet to an iron pin set; thence,
- 25. South 76° 28' 58" East, 666.51 feet to an iron pin set; thence,
- 26. South 08° 33' 32" West, 567.37 feet to an iron pin set; thence,
- 27. South 00° 41' 42" West, 87.99 feet to an iron pin set; thence,
- 28. South 02° 25' 18" West, 183.52 feet to an iron pin set; thence,
- 29. South 17° 26' 19" West, 174.68 feet to an iron pin set; thence,
- 30. South 10° 04' 17" West, 57.60 feet to an iron pin set; thence,
- 31. South 08° 42' 11" East, 30.88 feet to an iron pin set; thence,
- 32. South 07° 33' 11" West, 109.07 feet to an iron pin set; thence, South 04° 56' 58" West, 154.40 feet to an iron pin set; thence,
- 33. South 04° 56′ 58" West, 154.40 feet to an iron pin set; thence, 34. South 05° 47′ 55" West, 347.37 feet to an iron pin set; thence,
- 34. South 05° 47' 55" West, 347.37 feet to an iron pin set; thence, South 07° 17' 39" West, 212.75 feet to an iron pin set; thence,
- 36. North 85° 45' 22" West, 89.60 feet to an iron pin set; thence,
- 37. North 83° 35' 40" West, 627.42 feet to an iron pin set; thence,
- 38. North 82° 45' 56" West, 209.77 feet to an iron pin set; thence,
- 39. North 83° 43' 52" West, 812.23 feet to an iron pin set; thence,
- 40. South 06° 32' 25" West, 843.60 feet to an existing iron pin; thence,
- North 84° 17' 37" West, 367.92 feet to an existing iron pin; thence, North 83° 45' 31" West, 1213.32 feet to an existing iron pin; thence,
- 42. North 83° 45' 31" West, 1213.32 feet to an existing iron pin; thence, 43. North 83° 47' 41" West, 277.42 feet to an existing iron pin; thence,
- 44. North 05° 52' 17" East, 217.39 feet to an iron pin set; thence,
- 45. North 06° 40' 39" East, 771.14 feet to an iron pin set; thence,
- 46. North 01° 29' 48" East, 54.05 feet to an iron pin set; thence,
- 47. North 08° 40' 27" East, 58.75 feet to an existing iron pin; thence,

		n z	

- 25. North 06° 45' 08" East, 453.94 feet to an iron pin set; thence,
- 26. North 80° 51' 56" West, 549.57 feet to an existing iron pipe; thence,
- 27. North 07' 14' 47" East, 1033.99 feet to an existing iron pipe; thence,
- North 84° 40' 17" West, 51.08 feet to an existing iron pipe on the easterly right-of-way of I-65; thence with said easterly right-of-way,
- 29. North 16° 15' 45" East, 445.07 feet to an iron pin set; thence leaving said right-of-way,
- 30. South 82* 28' 23" East, 649.73 feet to an iron pin set; thence,
- 31. North 16' 15' 45" East, 539.80 feet to an iron pin set; thence,
- 32. South 82° 17' 10" East, 1173.81 feet to an iron pin set; thence,
- 33. South 83° 15' 38" East, 203.25 feet to a 42 inch Hackberry Tree; thence,
- 34. South 07° 22' 53" West, 1600.93 feet to the POINT OF BEGINNING and containing 314.9547 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Gary T. Baker, Trustee, of record in Book 1121, Page 345, Register's Office for Williamson County, Tennessee. The above described property also includes a cemetery lying on the westerly portion of the property.

Tract 215

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 215):

Beginning at iron pin set in the centerline of Liberty Pike, said point lying northerly along the centerline of Liberty Pike 869.93 feet from the intersection of the northerly right-of-way of Cool Springs Boulevard, also being the southeasterly corner of the tract described herein; thence leaving said centerline,

- 1. North 81° 57' 35" West, 584.38 feet to an existing iron pipe; thence,
- 2. North 06° 36' 05" East, 1131.99 feet to an iron pin set; thence,
- 3. North 81° 52' 43" West, 921.61 feet to an iron pin set in the easterly right-of-way of I-65; thence with said easterly right-of-way the following calls:
- 4. North 16° 14' 43" East, 356.82 feet to an existing concrete highway monument; thence,
- 5. North 28° 42' 16" East, 507.76 feet to an existing concrete highway monument; thence,
- 6. North 61° 14' 08" East, 282.97 feet to an existing concrete highway monument; thence,
- 7. North 16° 11' 55" East, 199.84 feet to an existing concrete highway monument; thence,
- 8. North 31° 27' 43" West, 296.95 feet to an existing concrete highway monument; thence,
- 9. North 07° 54' 15" West, 219.46 feet to an existing concrete highway monument; thence,
- 10. North 16° 12' 55" East, 858.06 feet to an existing concrete highway monument; thence leaving the easterly right-of-way of I-65,
- 11. South 84° 40' 17" East, 51.08 feet to an existing iron pipe; thence,

Beginning at iron pin set in the centerline of Liberty Road, said point lying 1807 feet southerly along the centerline of Liberty Road from the centerline of Moores Lane and being the northeasterly corner of the tract described herein; thence with the centerline of Liberty Road,

- 1. South 07° 06' 34" West, 205.85 feet to an iron pin set; thence leaving said centerline,
- 2. North 83° 35' 19" West, 447.25 feet to an iron pin set; thence,
- 3. South 06° 20' 45" West, 599.74 feet to an existing iron pipe; thence,
- 4. South 83° 35' 19" East, 439.25 feet to an iron pin set in the centerline of Liberty Road; thence with said centerline the following calls:
- 5. Southwesterly, 247.40 feet along the arc of a curve to the right having a radius of 235.42 feet, a central angle of 60° 12' 43", and a chord bearing and distance of South 37° 12' 56" West, 236.17 feet to an iron pin set; thence,
- 6. South 67° 19' 17" West, 130.57 feet to an iron pin set; thence,
- 7. Southwesterly, 673.13 feet along the arc of a curve to the left having a radius of 1378.10 feet, a central angle of 27° 59' 10", and a chord bearing and distance of South 53° 19' 42" West, 666.46 feet to an iron pin set; thence,
- 8. South 39° 20' 07" West, 76.13 feet to an iron pin set; thence leaving the centerline of Liberty Road,
- 9. North 07° 22' 53" East, 1600.93 feet to a 42 inch Hackberry Tree; thence,
- 10. South 82° 50' 47" East, 745.96 feet to the POINT OF BEGINNING and containing 14.5224 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from First Tennessee Bank National Association, of record in Book 1121, Page 338, Register's Office for Williamson County, Tennessee.

Tract 219

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 219):

Beginning at an existing iron pipe on the westerly right-of-way of Liberty Road, said point lying 2013 feet southerly along the westerly right-of-way of Liberty Road from the centerline of Moores Lane and being the northeasterly corner of the tract described herein; thence with the westerly right-of-way of Liberty Road,

- 1. South 06° 53' 59" West, 599.76 feet to an iron pin set; thence,
- 2. North 83° 35' 19" West, 426.50 feet to an existing iron pipe; thence,
- 3. North 06° 20' 45" East, 599.74 feet to an iron pin set; thence,

4. South 83° 35' 19" East, 432.30 feet to the POINT OF BEGINNING and containing 5.9120 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Gary T. Baker, Trustee, of record in Book 1121, Page 345, Register's Office for Williamson County, Tennessee.

Tract 220

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 220):

Beginning at iron pin set in the centerline of Liberty Road, said point lying 2127 feet southerly along the centerline of Liberty Road from the centerline of Moores Lane and being the northwesterly corner of the tract described herein; thence leaving the centerline of Liberty Road,

- 1. South 83° 02' 25" East, 523.60 feet to an existing iron pipe; thence,
- 2. South 06° 53' 29" West, 299.51 feet to an existing iron pipe; thence,
- 3. South 82° 18' 20" East, 1001.61 feet to an existing iron pipe; thence,
- 4. South 06° 58' 27" West, 1447.49 feet to an existing iron pipe; thence,
- 5. North 82* 26' 33" West, 2570.13 feet to an iron pin set in the centerline of Liberty Road; thence with said centerline the following calls:
- 6. North 39° 20' 07" East, 613.92 feet to an iron pin set; thence,
- 7. Northeasterly, 673.13 feet along the arc of a curve to the right having a radius of 1378.10 feet, a central angle of 27° 59' 10", and a chord bearing and distance of North 53° 19' 42" East, 666.46 feet to an iron pin set; thence,
- 8. North 67° 19' 17" East, 130.57 feet to an iron pin set; thence,
- 9. Northeasterly, 247.40 feet along the arc of a curve to the left having a radius of 235.42 feet, a central angle of 60° 12′ 43″, and a chord bearing and distance of North 37° 12′ 56″ East, 236.17 feet to an iron pin set; thence,
- 10. North 07° 06' 34" East, 486.10 feet to the POINT OF BEGINNING and containing 70.7919 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from First Tennessee Bank National Association, of record in Book 1121, Page 338, Register's Office for Williamson County, Tennessee.

Tract 221

A tract of land in the Eighth Civil District of Williamson County, in the City of Franklin, Tennessee, and being more particularly described as follows according to a survey by Ragan-Smith-Associates, dated September 23, 1993, Job No. 85-132, Work Order 4290 (Tract 221):

Beginning at iron pin set on the easterly right-of-way of Liberty Road, said point lying 876 feet southerly along the easterly right-of-way of Liberty Road from the centerline of Moores Lane and being the northwesterly corner of the tract described herein; thence leaving the easterly right-of-way of Liberty Road,

- 1. South 82° 36' 47" East, 1508.20 feet to an existing iron pipe; thence,
- 2. South 08' 37' 06" West, 91.81 feet to an existing iron pin; thence,
- 3. South 06° 34' 22" West, 587.97 feet to an existing iron pin; thence,
- 4. South 06° 52' 56" West, 542.88 feet to an iron pin set; thence,
- 5. South 07° 28' 51" West, 329.39 feet to an existing iron pipe; thence,
- 6. North 82° 18' 20" West, 1001.61 feet to an existing iron pipe; thence,
- 7. North 06° 53' 29" East, 299.51 feet to an existing iron pipe; thence,
- 8. North 83° 02' 25" West, 508.60 feet to an iron pin set on the easterly right-of-way of Liberty Road; thence with said easterly right-of-way,
- 9. North 07° 06' 34" East, 1250.87 feet to the POINT OF BEGINNING and containing 50.2157 acres, more or less as calculated by the above courses which were determined within the precision requirements of a "URBAN" Class ALTA/ACSM Land Title Survey of 1992.

Being part of the same property conveyed to Cool Springs Real Estate Associates, L.P., by deed from Gary T. Baker, Trustee, of record in Book 1121, Page 345, Register's Office for Williamson County, Tennessee.